

Delft University of Technology

Transportation technologies, sharing economy, and teleactivities Implications for built environment and travel

Mouratidis, Kostas; Peters, Sebastian; van Wee, Bert

DOI 10.1016/j.trd.2021.102716

Publication date 2021 **Document Version** Final published version

Published in Transportation Research Part D: Transport and Environment

Citation (APA) Mouratidis, K., Peters, S., & van Wee, B. (2021). Transportation technologies, sharing economy, and mouratidis, K., Peters, S., & van Wee, B. (2021). Transportation technologies, sharing economy, and travel. *Transportation Research Part D: Transport* teleactivities: Implications for built environment and travel. Transportation Research Part D: Transport and Environment, 92, Article 102716. https://doi.org/10.1016/j.trd.2021.102716

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect



Transportation Research Part D



journal homepage: www.elsevier.com/locate/trd

Transportation technologies, sharing economy, and teleactivities: Implications for built environment and travel



Kostas Mouratidis^{a,*}, Sebastian Peters^a, Bert van Wee^b

^a Department of Urban and Regional Planning, Faculty of Landscape and Society, Norwegian University of Life Sciences (NMBU), Ås, Norway ^b Transport and Logistics Group, Faculty of Technology, Policy and Management, Delft University of Technology, Delft, the Netherlands

ARTICLE INFO

Keywords: Emerging mobility Information and communications technology (ICT) Urban form Smart cities Literature review Coronavirus disease (COVID-19) pandemic

ABSTRACT

This paper reviews how teleactivities, the sharing economy, and emerging transportation technologies – components of what we could call the "App City" – may influence travel behavior and the built environment. Findings suggest that teleactivities may substitute some trips but generate others. Telework and teleconferencing may reduce total travel. Findings on the sharing economy suggest that accommodation sharing increases long-distance travel; bikesharing is conducive to more active travel and lower car use; carsharing may reduce private car use and ownership; ridesourcing (ridehailing) may increase vehicle miles traveled; while the implications of e-scooter sharing, ridesharing, and Mobility as a Service are context-dependent. Findings on emerging transportation technologies suggest that private autonomous vehicles and urban air mobility may increase total travel, whereas autonomous buses may lead to reduced car use. Implications of App Cities for the built environment include new transport systems and land use changes due to behavioral changes.

1. Introduction

The rapid developments in information and communications technology (ICT) and the use of mobile or computer online applications (apps) are enabling the widespread adoption of teleactivities, the exponential rise of the sharing economy, and the emergence of new transportation technologies (Gössling, 2018; Levinson & Krizek, 2017; Lyons, Mokhtarian, Dijst, & Böcker, 2018). Contemporary cities have started to accommodate, to different degrees, these three elements: teleactivities, sharing economy, and emerging transportation technologies. Teleactivities such as teleworking and online shopping are becoming mainstream in many societies. The Coronavirus disease (COVID-19) outbreak in 2019–2020 provided an additional strong boost to the widespread global application of teleactivities (Wijesooriya, Mishra, Brand, & Rubin, 2020). Shared mobility options such as sharing of bikes, electric scooters, or cars were making their appearance in more and more cities before the COVID-19 crisis occurred. Emerging transportation technologies including autonomous vehicles, drones, and robots are being tested or already being used in certain cases. Drones have also been used during COVID-19 for purposes such as surveillance and delivery of supplies.

The current dominant trends in transport and land use literature highlight the potential of these app-enabled developments for changing travel behavior and the built environment (Pawlak et al., 2019). Although research on these topics has been growing rapidly, a holistic understanding of how app-enabled teleactivities, the sharing economy, and emerging transportation technologies could potentially influence travel and the built environment is lacking in existing literature. A synthesis of relevant literature is important in

* Corresponding author.

E-mail address: konstantinos.mouratidis@nmbu.no (K. Mouratidis).

https://doi.org/10.1016/j.trd.2021.102716

Available online 5 February 2021 1361-9209/© 2021 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/hy/4.0/). order to gain insights into how human settlements and life in them are changing and how they are expected to change in the future. This knowledge will offer useful foundations for future research in these fields. It can also provide input to urban governance and planning aiming to proactively steer digitalization and new mobility options toward desired outcomes.

This paper attempts to address this gap by synthesizing the significant trends occurring in transportation and land use due to appenabled developments. It presents a synthesis of literature on the potential influences of teleactivities, the sharing economy, and emerging transportation technologies on travel behavior and the built environment. The paper examines two research questions: (1) What might be the implications of teleactivities, sharing economy, and emerging transportation technologies for travel behavior? (2) What might be the implications of teleactivities, sharing economy, and emerging transportation technologies for the built environment? Teleactivities reviewed in the paper are: telecommuting (telework), online shopping, online education, teleconferencing, teleleisure, telehealth, and online social networking. Sharing economy options reviewed in the paper are: accommodation sharing and shared mobility including carsharing, ridesharing, ridesourcing (ridehailing), bikesharing, e-scooter sharing, and Mobility as a Service. Emerging transportation technologies reviewed in the paper are: autonomous vehicles, delivery drones, delivery robots, and urban air mobility. Implications for travel behavior will be examined mainly in terms of changes in modal shares, travel distances, trip purposes, and trip frequencies. Implications for the built environment will be examined in terms of changes in location choices, land uses, and transport systems.

Our review includes a qualitative interpretation of existing evidence and theoretical reflections. The scope of the review is broad; therefore, we do not aim to provide an exhaustive review for each element covered in the paper, but rather a synthesis of literature (see Appendix for more details). Around 200 studies were reviewed. We provide an overview of the state of knowledge by presenting and critically interpreting empirical evidence, reflecting on causal mechanisms, reflecting on the research questions at a conceptual level when empirical evidence is missing, identifying gaps and limitations, and presenting recommendations for future research.

The synthesis of literature has led to the introduction of a new concept: the "App City", a concept we use as an umbrella term to describe the emerging trends that we have identified. To facilitate structuring the literature review, discussing the review as a whole, and enhancing readability, we introduce the App City concept at an early stage here. App City will be defined as the city where residents and visitors may use apps to perform teleactivities, participate in the sharing economy, and engage in emerging transportation technologies. These three components of App Cities have three common characteristics. They: (a) are enabled by the use of online apps by end-users (residents and visitors), (b) are strongly interconnected and influence each other, and (c) may significantly influence both the built environment and travel behavior. Other technological developments and ongoing digitalization also rapidly change the built environment and society, but we identify those three in particular as the most significant ones in relation to understanding ongoing and near-future changes in land use and transport. It also has to be acknowledged that the three components of App City are not only present in cities but also in smaller human settlements. However, we use the term App City for reasons of simplicity, and also because several of these developments are more prominent in cities (e.g. teleworking, bikesharing, ridesourcing, escooter sharing, Mobility as a Service) than in rural areas.

The paper is structured as follows. Section 2 presents basic theoretical background on teleactivities, the sharing economy, and emerging transportation technologies and a conceptual model that represents their links to travel behavior and the built environment. Sections 3–5 present a literature review of the implications of teleactivities, sharing economy, and emerging transportation technologies for travel behavior and the built environment. Section 3 focuses on teleactivities, Section 4 focuses on the sharing economy, and Section 5 focuses on emerging transportation technologies. Section 6 presents a summary of findings, a discussion, and an agenda for future research.

2. Background

The literature review is organized around three major areas: teleactivities, the sharing economy, and emerging transportation technologies. These three areas are based on the use of ICT and mobile or computer apps and are contributing to changes in people's travel behavior and transformations of the built environment (Banister & Stead, 2004; Batty, 2020; Gössling, 2018; Kwan, 2007; Levinson & Krizek, 2017; Line, Jain, & Lyons, 2011; Lyons et al., 2018; Newton, 2012; van Wee, Geurs, & Chorus, 2013; Yousefi & Dadashpoor, 2019).

Teleactivities, the sharing economy, and emerging transportation technologies – components of what we call "App City" in this paper – are all interconnected. Teleactivities are linked with shared mobility because sharing (cars, bikes, rides, e-scooters) is much easier with ICT-enabled teleactivities (information provision, reservations, paying, insuring) than without. And if shared mobility becomes the standard, this would further stimulate teleactivities, such as online shopping. For example, if people replace private car ownership with carsharing, for each trip the threshold to start driving becomes larger. Booking, walking to find the car, and paying may make each shared car trip slightly less convenient than personal car use, thereby triggering a shift towards teleactivities, such as online shopping, as opposed to driving to a shop. The sharing economy and emerging transportation technologies are also linked: autonomous vehicles, drones, robots, and urban air mobility options might be shared in the future and merged into what could be called "future shared mobility", because sharing reduces the costs of these mobility options. Smart mobility may facilitate tele-activities. For example, autonomous driving is linked with teleactivities since people are allowed to multitask while traveling – they are enabled to both travel and perform online activities (e.g. teleconferencing, teleleisure, telework) instead of simply driving. Accommodation sharing is facilitated by the teleactivity of online booking. In addition, if people would substitute activities for teleactivities, they might increase travel for other purposes, including recreation making use of accommodation sharing, as made explicit by the concept of constant travel time budgets – see Section 3.1.

Our literature review is based on the conceptual model as visualized in Fig. 1. The remainder of the paper will attempt to shed light

on the state of knowledge on teleactivities, the sharing economy, and emerging transportation technologies and their relation to travel behavior and the built environment. The knowledge provided in this paper aims to provide the background for future research and discussion on the potential societal and environmental outcomes of App Cities.

3. Teleactivities

Several activities that have traditionally occurred in the physical world are now occurring in the virtual world, enabled by ICT. Activities including working, shopping, learning, recreation, networking, socializing, and making new acquaintances are often happening online via ICT. Activities that occur remotely – nowadays usually via ICT – are called teleactivities. Teleactivity is not a synonym for online activity, but several online activities – e.g. online shopping and online education – are nowadays occurring remotely without being physically present in "brick and mortar" facilities. Therefore, online activities such as online shopping and online education can be considered teleactivities. The employment of teleactivities has been accelerated dramatically due to COVID-19 (Wijesooriya et al., 2020).

ICT-enabled teleactivities are allowing easier access to people, goods, and information. By allowing easier access to people, goods, and information, teleactivities in turn affect behavior and the built environment (Andreev, Salomon, & Pliskin, 2010; Yousefi & Dadashpoor, 2019). According to Circella and Mokhtarian (2017), ICT and teleactivities may influence travel and the built environment in several ways such as by influencing the location of residences and businesses, vehicle ownership and travel mode choices, and eventually the spatial forms of cities. ICT-enabled teleactivities also offer a wide range of options for multi-tasking, especially while traveling (Kenyon & Lyons, 2007; Pawlak, 2020). People are able to perform more than two activities at the same time; for example, travel and telework, travel and socialize online, and travel and perform educational or recreational activities.

The mechanisms through which teleactivities are linked to human behavior, and most notably travel behavior, are classified by previous studies (Andreev et al., 2010; Mokhtarian, Salomon, & Handy, 2006; Salomon, 1986) as: (a) substitution – replacement of a traditional activity by a virtual activity leading to a decrease in travel), (b) complementarity – a traditional activity is supplemented by a virtual one leading to an increase in travel, (c) modification – ICT changes the ways in which an activity occurs potentially affecting travel behavior, (d) neutrality – no change occurs in activities and travel due to ICT. The impacts of ICT-enabled teleactivities on travel behavior often include combinations of these mechanisms, and this makes them complex and difficult to quantify (Gössling, 2018).

Teleactivities may lead to changes in the built environment in two ways: by increasing accessibility to some activities and services thus enabling living in or relocating to more remote areas and favoring the expansion of urban areas; and by replacing activities that have been traditionally occurring in conventional stores and facilities thus reducing the need for such stores and facilities. More specifically, (a) teleactivities increase virtual accessibility, flexibility, and reduce geographical restrictions, and thus may provide opportunities for the relocation of households or workplaces to more remote areas (Moriset, 2003). Thereby, they could potentially be encouraging urban expansion and decentralization (Yousefi & Dadashpoor, 2019). However, it should be noted spatial development mostly depends not on the technologies themselves, but on political intentions, spatial planning policies, and how technologies will be utilized to reach certain goals (Lyons et al., 2018). (b) By providing virtual access to a wide range of activities and products, tele-activities may have a substitution effect on some traditional activities that occur in "brick and mortar" stores and facilities (Circella & Mokhtarian, 2017). Thereby, demand for such stores and facilities becomes lower and they eventually start to become scarcer or even disappear (Lyons et al., 2018).

The following subsections review different teleactivities – telecommuting, online shopping, online education, and other teleactivities including teleconferencing, teleleisure, telehealth, and online social networking – and how these relate to travel behavior and



Fig. 1. Conceptual model linking App City components to travel behavior and the built environment.

the built environment (see also Table 1 for an overview).

3.1. Telecommuting

Telecommuting – also called telework or teleworking – is the possibility of not traveling to the main workplace on a regular basis and working from another location (Pratt, 2000). Using digital devices and the internet, telecommuters may work from home but also in several other places such as co-working spaces, coffee shops, and libraries (Di Marino et al., 2018). Telecommuters may work mainly from distance or may have the flexibility to travel to work sporadically (Hill et al., 2003). Telecommuting has been expanding largely due to ICT and especially in larger urban regions (Vilhelmson & Thulin, 2016). COVID-19 made telecommuting a necessity for several job sectors, and this might lead to changes in attitudes towards flexible and remote working. The adoption of telecommuting depends on the nature of the job and the working arrangements and work culture, while it is associated with personal characteristics such as age and education (de Graaff & Rietveld, 2007). It also depends on managers' trust and control as well as individual and household work–life balance issues (Vilhelmson & Thulin, 2016). Telecommuting may have positive impacts on labor productivity, especially for workers with long commutes times (Kazekami, 2020). Telecommuting may have small but mainly beneficial effects on perceived

Table 1

Summary	of how	teleactivities	may influence	e travel	behavior	and	the bui	lt environment.
---------	--------	----------------	---------------	----------	----------	-----	---------	-----------------

Teleactivities	Travel behavior	Built environment	Source
Telecommuting / telework	Can reduce overall distance traveled for commutingBut may also induce additional non-work travel	 Appearance of new spaces and stores such as co-working spaces and telework- friendly coffeeshops Potential to allow residing or working in remote areas thus favoring decentralization or urban expansion, but this mainly depends on spatial policies 	Andreev et al. (2010); Choo et al. (2005); de Graaff and Rietveld (2007); He and Hu (2015); Helminen and Ristimäki (2007); Kim (2017); Kim et al. (2012); Ory and Mokhtarian (2006); Pérez Pérez (2004); Tayyaran and Khan (2003); Tayyaran et al. (2003); Yousefi and Dadashpoor (2019); Zhu (2012)
Online shopping/ e-shopping	 Does not fully replace store shopping Linked to additional shopping activity and more shopping trips May replace out-of-home lei- sure activities 	 Several types of "brick and mortar" stores and facilities are decreasing in number, are disappearing, or have disappeared (e. g. bookstores, travel agencies, bank branches) By offering increased accessibility, online shopping may allow residing in remote areas thus favoring decentralization or urban expansion, but this mainly depends on spatial policies 	Andreev et al. (2010); Cao et al. (2013); Cao et al. (2010); Cao et al. (2012); Circella and Mokhtarian (2017); Ding and Lu (2017); Farag et al. (2006); Ferrell (2004); Freathy and Calderwood (2013); Lee et al. (2017); Lyons et al. (2018); Mokhtarian (2004); Mokhtarian et al. (2006); Rotem- Mindali and Salomon (2007); Zhen et al. (2016); Zhen et al. (2018); Zhou and Wang (2014)
Online education	 Reduces travel for education May have complementary rebound effects on travel by saving time for other types of travel 	No clear implications yet	Herring and Roy (2002); Roy et al. (2008); Wang and Lindsey (2019)
Teleconferencing	• Reduces travel for meetings	• No clear implications yet	Denstadli (2004); Geitmann (2020); Guerin (2017); Høyer and Næss (2001)
Teleleisure	 May replace travel for certain leisure activities May have complementary rebound effects by saving time for other types of travel 	• Several types of "brick and mortar" stores and facilities are decreasing in number, are disappearing, or have disappeared (e. g. music stores, cinemas)	Aguiléra et al. (2012); Andreev et al. (2010); Mokhtarian et al. (2006); Wang and Law (2007)
Telehealth	 Reduces travel to access health services May have complementary rebound effects by saving time for other types of travel 	• No clear implications yet	Dorsey and Topol (2016); Holmner et al. (2014); and authors' elaborations
Online social networking	 May replace travel for certain social activities May have complementary rebound effects by saving time for other types of travel May generate opportunities for additional travel for face-to-face socializing 	• No clear implications yet	Jamal and Habib (2020); Rosenfeld et al. (2019); and authors' elaborations

autonomy and work-life balance, while it may improve outcomes including job satisfaction, performance, turnover intent, and role stress (Gajendran & Harrison, 2007). A risk with telecommuting is professional isolation which may negatively affect job performance (Golden et al., 2008).

Telecommuting may reduce work-related travel as compared to typical commute since the worker does not travel to the main workplace on a regular basis. Therefore, telecommuting has the potential to reduce overall travel demand, transport emissions, and air pollution, and to relieve transport infrastructure (Pérez Pérez, 2004). However, studies on telecommuting and travel behavior report mixed results. On the one hand, a study in Finland found that telecommuting is associated with reduced overall distance traveled for commuting (Helminen & Ristimäki, 2007), and a time-series study from the USA reported that telecommuting is linked to a reduction in annual total distance traveled – including work and non-work trips – of 0.8% or less (Choo et al., 2005). On the other hand, although there are several studies reporting substitution effects of telecommuting (Andreev et al., 2010), there are recent studies reporting complementary effects or non-significant effects. Zhu (2012) suggested that telecommuting has a complementary effect on travel, since it was found to be associated with travels longer in distance and duration both for total work trips and total non-work trips. Another study from Chicago, USA found that telecommuting was associated with a lower number of commute trips (as expected), but a higher number of non-work trips (He & Hu, 2015). A study from the Netherlands found no significant difference in the total commuting distance traveled between telecommuters and non-telecommuters (Gubins et al., 2019).

On the one hand, these mixed results could be attributed to the individual differences between telecommuters and nontelecommuters and not to a complementary effect of telecommuting on travel. First, telecommuters may travel longer distances to arrive at the main workplace (on days that they travel to their main workplace) than non-telecommuters. In Finland, for example, the commuting distance per trip was 3.9 km longer on average for telecommuters than that of non-telecommuters (Helminen & Ristimäki, 2007). A long commuting trip is one of the main motives for choosing to telecommute (Helminen & Ristimäki, 2007), while the opposite could also be the case: the possibility of telecommuting could increase the willingness to accept a longer commute (De Vos et al., 2019). Therefore, even if telecommuters skip some trips to work, their longer commute distance per trip may be responsible for balancing the difference for the total commuting distance between telecommuters and non-telecommuters. The total commuting distance covered by telecommuters will also depend on the definition of telecommuting status, how often they work remotely, and from where they work remotely. Second, if telecommuters are indeed inclined to make a higher number of non-work trips (He & Hu, 2015), due to personal attitudes and not telecommuting itself (He & Hu, 2015; Lyons et al., 2018), then their non-work travel distance would be longer than that of non-telecommuters, also adding up to the total distance traveled.

On the other hand, there might be indeed a complementary effect of telecommuting on travel. If people could save travel time because of ICT use, this does not mean they actually will do so. An influential theory explaining why is the theory of "constant travel time budgets" (Mokhtarian & Chen, 2004): measured over a large group of people, such as a country, the average time spent on travel is fairly constant at about 60–75 min per person per day. Less time needed for travel, because of ICT or a faster transport system, therefore does not lead to less time spent traveling. People may adapt their trip destination and sometimes also residential choices and trip frequencies. It is possible that telecommuting may have indirect impacts on travel via three mechanisms (Helminen & Ristimäki, 2007). First, by replacing some commute trips, it may provide time and flexibility for additional non-work trips such as trips for maintenance of the household, picking up kids, or other out-of-home activities (He & Hu, 2015; Kim, 2017). Second, by providing the possibility of less frequent commuting, telecommuting may encourage residential relocation to more remote areas (Tayyaran & Khan, 2003; Tayyaran et al., 2003) and the adoption of lifestyles that involve longer travel distances (Zhu, 2012). A third option is that people change their workplace. For example, they might accept a job further away from their residential location because they have to travel to work less frequently.

Therefore, although telecommuting has the potential to reduce travel distances for commuting (Andreev et al., 2010), it seems that at the same time it may generate additional vehicle miles traveled by encouraging more non-work trips or the relocation to more remote areas. Comparing telecommuters versus non-telecommuters, as done by most existing studies, (instead of studying the same individuals before and after adopting telecommuting) makes it difficult to disentangle the possible effects of the above-mentioned mechanisms. In addition, all these various mechanisms may play out differently in different contexts with different working cultures, work regulations, or societal norms.

The impact of telecommuting on the built environment takes mainly two forms: firstly, emerging spaces and working environments for telecommuters and, secondly, possible expansion of urban areas. Telecommuting enables people to work remotely by increasing virtual accessibility and flexibility and reducing geographical restrictions (Moriset, 2003), thus potentially encouraging urban expansion and decentralization (Yousefi & Dadashpoor, 2019). New types of spaces have emerged to accommodate this demand such as co-working spaces or telework-friendly coffee shops. Telecommuters are more likely to live on the outskirts of cities or in remote areas (Kim et al., 2012). Some scholars have argued that telecommuting could facilitate living further away from work and this could therefore encourage urban expansion (Tayyaran & Khan, 2003; Tayyaran et al., 2003), while others reject the possibility of such a causal relationship (Kim et al., 2012; Ory & Mokhtarian, 2006). Although it seems that telecommuting, and ICT in general, may provide opportunities for the relocation of households or workplaces to more remote areas, spatial development mainly depends on political intentions, spatial planning policies, and how technologies will be utilized to reach certain goals (Lyons et al., 2018).

3.2. Online shopping

Online shopping – also called e-shopping – is a type of electronic commerce where one can buy products from a seller over the internet. Online shopping popularity is increasing rapidly, and it now occupies a large share of the total commercial activity in many countries. Online shopping is conducive to the "dematerialization" of certain categories of goods and the emergence of "bricks and

clicks" forms of businesses (Circella & Mokhtarian, 2017). A wide range of products are nowadays bought online replacing conventional commerce: newspapers, music, games, electronics, furniture, clothes, meals, groceries, vacation packages, tickets, and bank products (Freathy & Calderwood, 2013; Pawlak et al., 2019). Online shopping is also combined with app-based ridehailing-type services in the form of on-demand, app-based delivery of goods, groceries, and meals (Li et al., 2020; Xi et al., 2020). This type of online shopping is becoming central to life in several urban areas and has been boosted during COVID-19 (Alaimo et al., 2020). Online shopping increases options for shopping, possibly inducing additional shopping activity. Online shoppers tend to be more frequently residing in urban areas since urban residents often have higher computer literacy and greater access to the internet (Cao et al., 2013). However, online shopping is especially important for residents in locations with low accessibility to stores such as rural areas, islands, and exurban areas (Cao et al., 2013; Freathy & Calderwood, 2013).

Online shopping appears to have a complementary effect on in-store shopping according to a review of relevant studies (Andreev et al., 2010). This means that, for the most part, online shopping does not substitute traditional shopping, but induces additional shopping activity by offering a wide range of shopping possibilities at great convenience and generates additional trips to "brick and mortar" shops for physically checking products before buying (Mokhtarian, 2004; Rotem-Mindali & Salomon, 2007). Online buyers are found to make more trips to "brick and mortar" shops than non-online buyers (Cao et al., 2010; Cao et al., 2012; Ding & Lu, 2017; Farag et al., 2006; Ferrell, 2004; Lee et al., 2017; et al., 2016; Zhou & Wang, 2014). These results hold also when accounting for confounding factors such as shopping attitudes (Cao et al., 2010; Lee et al., 2017; Zhen et al., 2016), however further research is necessary to claim a causal link from online shopping to in-store shopping trips and fully understand the pathways between them (Lee et al., 2017). Contrary to most previous studies, Xi et al. (2018) find that store shopping may increase online shopping and not the other way around. Another link between online shopping and travel behavior was found to be a negative association between online buying and the frequency of leisure activities (Ding & Lu, 2017), suggesting that online shopping may replace traditional leisure activities and out-of-home trips (Ding & Lu, 2017; Mokhtarian et al., 2006).

Online shopping, combined with the digitalization of a wide range of products, may lead to changes in the built environment and specifically in commercial land uses. Several types of "brick and mortar" stores are likely to become scarcer due to online shopping taking over their market share (Circella & Mokhtarian, 2017; Lyons et al., 2018). Examples of such stores include music stores, software-selling stores, travel agencies, bookstores, video clubs, and bank branches. The impact of online shopping on transport infrastructure has not been explored much. For the moment, it seems to be small but it has been suggested that it may grow in the future (Zhen et al., 2016). The increased demand for online shopping may lead to more delivery vehicles in the central parts of cities, possibly resulting in increased traffic congestion and emissions if unregulated (World Economic Forum, 2020). To get a better understanding of how online shopping may influence travel, analysis of emerging travel patterns due to online shopping should include growing and more complex patterns of delivery traffic. Online shopping can improve shopping accessibility in low-density areas where conventional, "brick and mortar" stores are less accessible (Zhen et al., 2018), so it could, in theory, favor urban expansion and decentralization. But as noted for telecommuting, spatial structure mostly depends on policy goals and spatial policies rather than the technology itself.

3.3. Online education

Online education is a form of education in which students may partially or completely replace physical presence at an educational facility with learning activities via the internet. Similar terms include distance learning, tele-education, distance education, online learning, and e-learning. Online education can be formal or informal. Formal online education involves formal online courses or even degrees offered by educational institutions, while informal online education may involve self-learning using resources from the internet such as videos, e-books, articles, and wiki sources. A combination of traditional and online education, called blended learning, is common nowadays (Xanthidis et al., 2016). Online education can provide easier access to learning as it usually involves lower costs and flexible schedules and environments. Online education thus provides access to learning for diverse groups of the population including poorer people in places where education is not free, people with mobility difficulties, people residing in remote places, people who live in places with limited or no educational opportunities, people who do not want to or cannot relocate to study, people who work or have other commitments, and parents with young children. Online education is also an important tool for continuous learning throughout professional life (Harun, 2001). According to some studies, attending courses online seems to result in similar learning outcomes in terms of skills and grades compared with traditional classroom learning at least for certain topics (McCutcheon et al., 2015; Summers et al., 2005). However, other studies point to lower outcomes for online learning students (Hurlbut, 2018; Xu & Jaggars, 2013). Satisfaction with the course may be lower for online students than for traditional classroom students (Summers et al., 2005), possibly due to the different sense of social presence and connectedness with other students (Bulu, 2012). Lack of computer skills or lack of access to ICT are barriers to online education. Moreover, although a wide range of topics can be taught online, there are courses that require a physical presence in the form of lab work or clinical work.

Online education reduces or eliminates travel to classrooms, and therefore seems to have a substitution effect on travel. In fact, studies have shown that courses taught online involve significant reductions in personal travel and relevant transport emissions compared with classroom-based courses (Herring & Roy, 2002; Roy et al., 2008). It is possible however that online education may also have rebound effects on travel (Wang & Law, 2007). Time or money saved with online education may be spent on traveling for other purposes including work, household duties, and leisure. Students chose a more remote school and continue to reside at their existing location, or they may choose not to move to a residential location near their school because of online education opportunities. Such potential complementary effects of online education on personal travel have been underexplored in existing research.

To our knowledge, no significant impact of online education on the built environment has been reported in academic research.

Online education could potentially result in changes in transport infrastructure and land use. Reductions in travel to the classroom and changes in travel patterns throughout the day due to rebound effects could result in changing transport demands. Increasing demands for online education could also lead to decreasing demands for school building size. However, all these impacts are still relatively small. By allowing people to study without changing their place of residence and relocating closer to education facilities, online education could potentially influence land use and spatial development.

3.4. Other teleactivities (teleconferencing, teleleisure, telehealth, online social networking)

Other teleactivities include teleconferencing, teleleisure, telehealth, and online social networking. Teleconferencing is the live exchange of information and live communication between two or more people from distance. ICT enables teleconferencing with various platforms of videoconferencing via the internet. ICT-enabled teleconferencing replaces in-person meetings with online meetings, and also allows the online delivery of seminars and remote attendance at scientific conferences. This was particularly highlighted during COVID-19 travel restrictions (Geitmann, 2020). Teleleisure is ICT-enabled leisure. Teleleisure activities replace traditional leisure activities. Examples of teleleisure activities include online gaming, online television, online streaming of movies and series, and listening to music online (Lobato, 2019). Telehealth or telemedicine is the ICT-enabled access to services and information related to health. Examples include video consultation between patient and doctor, online psychotherapy, video conference between clinicians, and online exchange of test results (Dorsey & Topol, 2016). The accelerated adoption of telehealth, which has been boosted by COVID-19, has shown that telehealth, especially if used proactively rather than reactively, may generate important health benefits, assist with the everyday and emergency challenges in healthcare, and reduce socio-spatial inequities related to health (Dorsey & Topol, 2016; Smith et al., 2020). Online social networking is the development and maintenance of social relationships and social networks via platforms on the internet. Online social networking can be used for linking professionals, for socializing with or finding existing friends and acquaintances, for making new friends, for finding a new partner or intimate relationship, or for communicating with people who share common interests or ideas. Socializing via apps and websites has been replacing traditional ways of meeting people (Rosenfeld et al., 2019). It may have positive implications for well-being when used carefully but may also lead to addictions, stress, and distraction (Tarafdar et al., 2020; Wenninger et al., 2019).

Teleconferencing replaces face-to-face meetings and can thus eliminate trips to other locations. Teleconferencing could result in a reduction in car and air travel (Denstadli, 2004; Guerin, 2017). With the widespread development of ICT, travel to conferences can now be replaced by online activities such as online education and information sharing, teleconferencing, and online social networking (Geitmann, 2020; Høyer & Næss, 2001). Similarly, telehealth in the form of online meetings with the doctor can replace face-to-face appointments and thus reduce personal travel and related carbon emissions (Holmner et al., 2014). Teleleisure replaces some leisure activities that were typically performed out of home with in-home leisure (Varghese & Jana, 2019). The time saved through ICT use for teleleisure and possibly telehealth may encourage additional personal travel, thus having a complementary effect on travel (Aguiléra et al., 2012; Andreev et al., 2010; Mokhtarian et al., 2006; Wang & Law, 2007). Online social networking may replace travel for some traditional social activities but could also have a complementary effect on travel (Jamal & Habib, 2020) by generating opportunities for additional travel for face-to-face socializing. Although teleleisure, telehealth, and online social networking may have complementary effects on travel, teleconferencing during work probably has a substitution effect since it is a mandatory activity. Travel for meeting people for work purposes is likely to be replaced by work at the workplace or even by telework. These typically require travel of shorter distance or no travel at all. Overall, the relationships between these teleactivities and travel have been understudied and are difficult to quantify (Andreev et al., 2010; Ettema, 2018; Gössling, 2018).

Similarly to online shopping, teleleisure reduces the use of "brick and mortar" shops and facilities. For example, online streaming of movies and services eventually may reduce visits to movie theaters and video clubs, resulting in a reduction of movie theaters and the disappearance of video clubs. Similar replacement impacts are observed for listening to music online and record shops. The other teleactivities such as teleconferencing, telehealth, and online social networking have not been linked to significant changes in land uses or transport infrastructure, although, theoretically, the increased accessibility that they offer may encourage residing in remote locations and thus favor decentralization and urban sprawl.

4. Sharing economy

Humans have been sharing materials and services with others long before ICT-based sharing options became available. However, in recent years, the sharing economy has been developing dramatically due to the widespread use of ICT (Botsman & Rogers, 2010; Gansky, 2010; Harris & Gorenflo, 2012). The current trends in the sharing economy have been criticized for losing the sense of actual "sharing" because of the mediating, for-profit role of companies such as Uber and Airbnb (Eckhardt & Bardhi, 2015). Bicycles, cars, car trips, electric scooters, and homes are now shared or, more accurately, rented between users and suppliers. In many cases, the supplier is a private company that owns a fleet of vehicles, homes, bikes, or e-scooters and rents them with the help of online apps. This business model resembles the business models of more traditional rental companies, contrasting with sharing between individuals. Therefore, whether this business model should be considered a form of the sharing economy is open to debate. In this review, we include all different types of business models under the category of "sharing economy" for reasons of simplicity.

Shared mobility (such as bikesharing, carsharing, ridesharing, e-scooter sharing) and accommodation sharing (e.g. Airbnb) are among the most commonly used forms of sharing economy (Dogru et al., 2020; Machado et al., 2018; Standing et al., 2019). In addition to shared mobility and accommodation sharing, there are also other forms of the sharing economy such as garden sharing or task sharing. These are not examined in the paper as they have smaller impacts on travel behavior and the built environment.

Table 2

Summary of how the sharing economy may influence travel behavior and the built environment.

Sharing economy	Travel behavior	Built environment	Source
Accommodation sharing	 By displacing residents to less central locations, it could induce increased travel distances May induce additional long-distance travel for tourism and business Extra income provided by rental accommodation sharing could incentivize those who rent their room or dwelling to engage in more frequent long-distance trips or holidays to destinations that are further away 	 Changes dwelling and building uses May result in changes in transport infrastructure and facilities and services New buildings for accommodation sharing purposes resulting in densification or urban expansion Induced long-distance travelers in the city by accommodation sharing will also require the expansion of existing transport infrastructure and addi- tional facilities and services 	Lyons et al. (2018); Tussyadiah and Pesonen (2015); and authors' elaborations
Carsharing	 Could reduce private car use and private car ownership Could reduce vehicle miles traveled If not intended to replace regular use of private cars, it could lead to increased walking, cycling, and use of public transport 	 Could reduce space for car parking Could free up urban space by reducing demands for car driving and parking New land uses may replace infrastructure for private cars Infrastructure for walking, cycling, and public transport may need to be expanded due to increased needs 	Kent (2014); Kent and Dowling (2013, 2016); Litman (2000); Martin and Shaheen (2011); Martin et al. (2010); Rotaris and Danielis (2018); Sioui et al. (2013); Stillwater et al. (2009)
Ridesharing (carpooling, vanpooling)	 May increase the mobility of low-income workers, singles, and women Mainly replaces the use of public transport Increases occupancy rate in cars and provides an alternative option to owning a second car May lead to lower demand for public transport that would, in turn, generate more car use and car ownership 	• No clear implications yet	Chan and Shaheen (2012); Lyons et al. (2018); Shaheen et al., 2016a; Shaheen and Cohen (2019); Shaheen et al. (2019)
Ridesourcing / ridehailing	 Replaces traditional taxi, but also public transport and the private car May induce additional travel and result in additional vehicle miles traveled 	 Could result in more cars traveling in the streets and increased congestion Could reduce car parking demand by replacing personal driving 	Alemi et al. (2018); Brown (2020); Erhardt et al. (2019); Henao and Marshall (2019a, 2019b); Jiao et al. (2020); Jin et al. (2018); Mohamed et al. (2019); Rayle et al. (2016); Shaheen and Cohen (2019); Tirachini (2020); Tirachini and del Río (2019)
Bikesharing	 Can increase cycling modal share Can replace use of public transport. walking, and car use Increases total active travel 	 Requires bikesharing infrastructure, conventional bicycle infrastructure, mixed-use built environment Can lead to reduced congestion and free space from reduced car use 	Castro et al. (2019); Duran-Rodas et al. (2019); Fishman (2016); Fishman and Cherry (2016); Fishman et al. (2013, 2014, 2015); Fyhri and Fearnley (2015); Guidon et al. (2019); Jia and Fu (2019); Shaheen et al. (2010); Wang and Lindsey (2019); Zhuang et al. (2019)
E-scooter sharing	 In compact, walkable, cyclable, and transit-oriented contexts: e-scooters mainly replace active travel and public transport In more car-oriented contexts: e-scooters seem to replace a high percentage of car travel, but also active travel and public transport Can be used to complement public transport and active travel modes as a "first-last mile" mobility option 	 E-scooters are now integrated into the built environment as they can be found on sidewalks, streets, and other public spaces If used to facilitate public transport and active travel and reduce car use, e-scooters could contribute to freeing up spaces occupied by cars and car infrastructure 	Fearnley et al. (2020); Gössling (2020); Hollingsworth et al. (2019); James et al. (2019); McKenzie (2019)
Mobility as a Service (MaaS)	• Potential for reduction in private car ownership and car use if	 If goals and policies aim for sustainable mobility, MaaS can contribute to a reduction in street 	Hensher (2017); Moscholidou and Pangbourne (2019); Pangbourne et al. (2019); Sochor et al. (2018); Utriainen

⁽continued on next page)

Table 2 (continued)

Sharing economy	Travel behavior	Built environment	Source	
	accompanied by appropriate goals and policiesRisk of increased car use and traffic if unregulated	 space for cars and reduction in space used for car parking, freeing up space for other land uses, and increasing demands for infrastructure for walking, cycling, and public transport. In that case, MaaS could be in line with urbanization and compact city policies. If unregulated, it could favor decentralization and urban expansion by facilitating mobility in more remote areas. 	and Pöllänen (2018); Wong et al. (2019)	

Accommodation sharing – also called home sharing – is the shared use of different types of accommodation. It is nowadays enabled by online platforms like Airbnb. Accommodation sharing may influence how people travel but it mainly does so on a larger scale. By offering an alternative, and at times cheaper, form of short-term accommodation, accommodation sharing may induce extra travel for tourism or business purposes, thereby having a complementary effect on total travel (Tussyadiah & Pesonen, 2015). Accommodation sharing is also changing the housing domain. Short-term rental of empty rooms or apartments can result either in higher building occupancy or in empty housing during low tourist season when dwellings are used only for tourism and not for living. On the other hand, as tourism expands, new buildings are being developed so that they can be used for accommodation sharing purposes.

Shared mobility is the shared use of a transportation mode that "enables users to gain short-term access to transportation modes on an as-needed basis" (Shaheen et al., 2016b, p. 77). Shared mobility is nowadays facilitated by ICT and apps (Gössling, 2018). It includes various forms of carsharing, bikesharing, ridesharing (carpooling and vanpooling), ridesourcing (on-demand ride services), and e-scooter sharing. Another concept that is relevant to shared mobility is that of "smart mobility". Smart mobility mainly refers to urban mobility that is enabled by ICT (Battarra et al., 2018; Lyons, 2018; Uteng et al., 2020). Therefore, smart mobility includes ICT-enabled shared mobility such as bikesharing and carsharing, but also emerging transportation technologies (reviewed in Section 5). Shared mobility may influence travel behavior in similar ways as teleactivities: substitution, complementary, modification, and neutrality. Moreover, by increasing accessibility and altering mobility options, shared mobility could potentially influence transport systems, land use, and location choices.

The following subsections review accommodation sharing and shared mobility, and how these relate to travel behavior and the built environment (see also Table 2 for an overview). The shared mobility subsection is divided into further subsections: carsharing, ridesharing, ridesourcing, bikesharing, e-scooter sharing, and Mobility as a Service.

4.1. Accommodation sharing

Accommodation sharing or home sharing is the concept of offering short-term stays in homes via online platforms. Accommodation sharing platforms can be classified into free, reciprocal, and rental (Voytenko Palgan et al., 2017). The most popular platform nowadays is offered by the short-term rental organization Airbnb. Accommodation sharing has been increasingly used (Dogru et al., 2020), and has consequently received praise but also fierce criticism. Positive aspects of accommodation sharing could be considered the potential for more efficient use of buildings, the extra income offered to residents who are renting their spaces, the exchange of cultures and experiences, the offer of a different type of accommodation experience, and the regeneration of buildings or neighborhoods in decline (Balampanidis et al., 2019; Guttentag, 2019). The other side of the coin, however, points to dramatic increases in rents especially in the case of very touristic cities and neighborhoods, displacement of residents and especially those of lower income, limited availability of dwellings for long-term rent when a short-term rental is more profitable, unfair competition with hotel businesses, nuisance for neighbors, and disruption of local social cohesion (Gurran & Phibbs, 2017; Guttentag, 2019). To mitigate the negative implications of accommodation sharing, several cities are forcing regulations and restrictions.

Although accommodation sharing does not appear to have direct impacts on the travel behavior of residents at a local level, it may influence local travel behavior indirectly. By displacing residents to less central locations, it could lead to increased travel distances at a city level (Lyons et al., 2018). Accommodation sharing also seems to have impacts on long-distance travel. It may induce additional long-distance travel for tourism and business (Tussyadiah & Pesonen, 2015). Accommodation sharing contributes to increased long-distance travel by offering more attractive prices, by covering alternative accommodation needs and sometimes a more personal and interactive experience, and by offering a wider variety of locations and accommodation types (Tussyadiah & Pesonen, 2015). Moreover, the extra income provided by rental accommodation sharing could incentivize those who rent their room or dwelling to engage in more frequent long-distance trips or holidays in destinations that are further away (Lyons et al., 2018).

Accommodation sharing could have the following implications for the urban built environment. First, for the case when the one who rents out a room or dwelling is the one who resides in the dwelling, accommodation sharing results in more efficient dwelling and building use. This would increase demands in transport infrastructure and local facilities and services, potentially inducing the

expansion of transport infrastructure and the appearance of extra facilities and services. Second, when dwellings are used only for tourism and not for living, accommodation sharing may result in empty housing during the low tourist season. Third, new buildings may be developed for accommodation sharing purposes resulting in densification or urban expansion. Fourth, accommodation sharing may lead to changes in residential choice and land use, since residents may choose or be forced to relocate. Fifth, long-distance travelers who visit the city (partially) attracted by accommodation sharing will also induce the expansion of existing transport infrastructure and the appearance of new facilities and services.

4.2. Shared mobility

4.2.1. Carsharing

Carsharing is a mobility option that allows the rental of shared cars, thereby having the potential to substitute use and ownership of private cars. The renting body could be a business, individual (peer-to-peer carsharing), or cooperative (Hampshire & Gaites, 2011). The aim of carsharing is that vehicles can be easily found parked in residential neighborhoods. They can nowadays be accessed using an app. Because of the easy access provided by smartphones and mobile internet (Ferrero et al., 2018), the use of carsharing is increasing and expanding, and thus carsharing is becoming a mainstream transport mode (Shaheen & Cohen, 2013). Due to its lower fixed costs, carsharing is a viable alternative to sporadic car users residing in urban areas, and this may incentivize urban residents to use other travel modes to a larger extent and reduce the use of the car (Litman, 2000). Carsharing could be integrated with Mobility as a Service and vehicle automation (Shaheen et al., 2019).

By providing an alternative to the private car, carsharing can reduce private car use (Sioui et al., 2013) and private car ownership (Kent, 2014; Martin & Shaheen, 2011; Martin et al., 2010). Due to the reduced private car ownership, carsharing may promote the use of environmentally friendly transport modes. Indeed, a review of relevant studies shows that carsharing can increase walking and cycling, and promote the use of public transport, while it may reduce vehicle miles traveled (Kent, 2014). For carsharing to promote sustainable mobility, it should be just complementary to more environmentally friendly travel modes such as walking, cycling, and public transport. It should not be used as a main transport mode, except for special cases. To be successful in replacing private car ownership and contribute to sustainable mobility, carsharing typically requires a dense, mixed-use built environment where walking, cycling, and public transport options are readily available (Kent & Dowling, 2013). However, there could also be a demand for carsharing in small towns and rural areas where private car ownership is usually higher, parking is more readily available and other transport options are limited (Rotaris & Danielis, 2018). The use of carsharing in small towns and rural areas was found to be more frequent among students and unemployed people, groups that may not have access to a private vehicle (Rotaris & Danielis, 2018).

Carsharing may cause changes in the built environment but these are not yet completely understood (Stillwater et al., 2009). Although parking spaces for carsharing are required (Kent & Dowling, 2013), overall carsharing can help reducing parking space required (Kent & Dowling, 2016) due to the reduced ownership of private cars (Kent, 2014; Martin & Shaheen, 2011; Martin et al., 2010). Reduced car use and increases in walking, cycling, and use of public transport (Kent, 2014; Sioui et al., 2013) can lead to reduced congestion and freed up spaces previously used for car travel and parking. Increased use of active travel and public transport may require the expansion of city infrastructure for walking, cycling, and public transport. Free space provided by a reduction in parking spaces and lower needs in road capacity for cars, could be used for other land uses including open public space as well as residential and commercial land uses.

4.2.2. Ridesharing

Ridesharing is a concept that includes carpooling and vanpooling (Chan & Shaheen, 2012), and involves sharing a car or van with other riders who travel to a similar destination (Shaheen & Cohen, 2019). It should not be confused with ridesourcing (e.g. Uber), which is an informal form of taxi where a passenger hires a personal car and driver to travel to a destination. Ridesharing has been traditionally done informally through social networks. Informal ridesharing examples are: a couple sharing a car to travel to work or two neighbors sharing a ride. ICT developments are now linked to the use of ridesharing. Ridesharing can now be performed between individuals who had not previously known each other using online platforms.

The exact impacts of ridesharing on travel behavior in cities are difficult to examine and research evidence remains inconclusive. Shaheen and Cohen (2019) reviewed existing studies and found that ridesharing may increase the mobility of groups with lower access to cars including low-income workers, singles, and women. In addition, they conclude that by increasing vehicle occupancy, ridesharing may also lead to reductions in vehicle miles traveled, energy consumption and emissions, and parking requirements. However, a study on carpooling found that most carpool riders were previous users of public transport (Shaheen et al., 2016a). This suggests that ridesharing mainly replaces the use of public transport. Therefore, ridesharing seems to increase the occupancy rate in cars and provide an alternative option to owning a second car (Lyons et al., 2018), but it could have a negative rebound effect on the use of public transport. This would lead to lower public transport demand that would in turn generate more car use and car ownership. To contribute to reduced total travel distances and reduced emissions, it seems that ridesharing needs to be used sporadically or solely when driving is the only option. Therefore, to contribute to sustainable mobility, it needs to be used in combination with regulations restricting car use, while promoting walking, cycling, and public transport.

4.2.3. Ridesourcing

Ridesourcing or ridehailing is an app-based, on-demand ride service such as Uber and Lyft (Rayle et al., 2016). Pooled variants of ridesourcing include ridesplitting, taxi sharing, and microtransit (Shaheen & Cohen, 2019). Residents of dense, mixed land use, and lower-income neighborhoods tend to use ridesharing more than residents of other types of neighborhoods, based on research in

California, USA (Alemi et al., 2018; Brown, 2020). Predictors of ridesourcing use may include technology-embracement, pro-environmental attitudes, variety-seeking attitudes, high education, frequent long-distance travel, and previous taxi and carsharing usage (Alemi et al., 2018). Pooled ridesourcing services may increase in the future as ridesourcing companies aim to increase their market shares and profits (Sperling, 2018). To increase sharing in ridesourcing services, (Brown, 2020) suggests that companies and cities should attract non-users, by offering improved pricing, among others. Societal benefits of ridehailing include comfort and security of passengers, increased mobility for car-free households and people with physical and cognitive limitations, and efficiency in riderdriver matching (Tirachini, 2020).

Ridesourcing replaces traditional taxi, but also public transport and the private car (Rayle et al., 2016; Tirachini & del Río, 2019; Young & Farber, 2019). Ridesourcing replacing public transport and active travel constitutes a risk for sustainable mobility (Brown, 2020). Ridesourcing usage is associated with a higher amount of trips (Jiao et al., 2020). This finding could be explained by the fact ridesourcing increases mobility due to the combination of on-demand service with lower costs. However, the finding could also be attributed to the characteristics of the ridesourcing users: technology-inclined individuals who tend to make more trips. A study from Denver, USA found that ridesourcing adds a significant amount of vehicle miles traveled to the system when accounting for traveling without a passenger (deadheading), induced travel, and substitution of more sustainable travel modes (Henao & Marshall, 2019a). A systematic review of literature indicates that ridesourcing should not be expected to reduce the ownership of private cars and that environmental impacts are still unclear (Jin et al., 2018). Findings from another literature review suggest that ridesourcing's substitution effect on public transport has increased motorized traffic and congestion, with possible negative impacts on the environment and energy consumption (Tirachini, 2020). Ridesourcing companies, such as Uber and Lyft, were found to the most important contributor to the increasing traffic congestion in San Francisco, USA (Erhardt et al., 2019). A positive environmental aspect of ridesourcing is that by replacing personal driving, it may reduce car parking demand in cities (Henao & Marshall, 2019b; Tirachini, 2020), freeing up space for other land uses. Policies and regulatory frameworks are expected to determine the influence of ridesourcing on travel and the built environment (Mohamed et al., 2019).

4.2.4. Bikesharing

Bikesharing aims to offer easy and affordable access to shared bicycles within urban areas. Bikeshare programs provide several benefits including flexible mobility, reduced greenhouse gas emissions, affordability, reduced congestion, reduced air pollution, health benefits from active travel, and a solution to the "first-last mile" issue associated with the use of public transport (Shaheen et al., 2010). The widespread use of ICT now allows the use of automated bikesharing stations as well as dockless bikesharing systems through mobile apps. Bikesharing has grown rapidly in recent years and more and more cities worldwide have been employing bikesharing systems (Fishman, 2016). Bikesharing has become a mainstream travel mode in several cities where residents use it to travel to work or to leisure activities (Fishman et al., 2013).

Empirical studies on bikesharing and travel behavior show that bikesharing has a substitution effect on public transport and walking as it mostly replaces trips previously made by these travel modes (Fishman, 2016; Fishman et al., 2013). At the same time, the adoption of bikesharing has been found to reduce car use – but to a smaller degree compared to the reduction in the use of public transport and walking (Fishman et al., 2013; Fishman et al., 2014). The adoption of dockless bikesharing has been linked to an increase in the proportion of cyclists and further reductions in car use (Jia & Fu, 2019). Overall, bikesharing increases levels of active travel by substituting sedentary modes of transport (Fishman et al., 2015). Bikesharing was found not to be associated with the number of trips made (Jiao et al., 2020) suggesting that it mainly has a substituting effect.

Using electric bicycles (e-bikes) in bicycle sharing programs is a new addition to urban transportation with the potential to change travel behavior in cities (Guidon et al., 2019). E-bike sharing might be preferable in specific topographies, for specific individuals, and for longer trips. E-bike sharing may be a new mobility option for older adults and other groups that would otherwise not have chosen biking as an option. It offers improved mobility to those not being able to bike without a motor. Electric bicycles can increase the amount of cycling by replacing sedentary transport modes (Fishman & Cherry, 2016; Fyhri & Fearnley, 2015). Using e-bikes can lead to important increases in physical activity when replacing car use or public transport, while net losses in physical activity for those switching from conventional cycling are usually small due to increases in overall travel distance (Castro et al., 2019).

Successful bikesharing programs require a dense, mixed-use built environment with high access to facilities and services, public spaces, and tourist attractions (Duran-Rodas et al., 2019; Wang & Lindsey, 2019). Efficient bikesharing infrastructure (e.g. docks), management, and maintenance together with an expansion in conventional cycling infrastructure (e.g. cycle lanes, bike parking) are also necessary to respond to increases in bike transport demand (Guidon et al., 2019; Shaheen et al., 2010; Zhuang et al., 2019). Increased use of bicycles and reductions in car use can lead to reduced congestion from car traffic. Some new land uses may be generated from this substitution effect which would be connected to a reduced need for road capacity and car parking.

4.2.5. E-scooter sharing

E-scooter sharing is the short-term rental of electric scooters (e-scooters) in cities. E-scooters are typically dockless and they are more commonly found in denser cities or denser parts of a city (Jiao & Bai, 2020), within geo-fenced urban areas. E-scooters are accessed with a smartphone app. E-scooter sharing made a sudden appearance in cities worldwide in 2017, and since then, the use of scooters has rapidly increased. They are considered a first- and last-mile mobility option and one of the most common forms of micro-mobility (McKenzie, 2019). For distances that are not too long, they can be used as a main travel mode instead of a "first-last mile" mode. E-scooters are associated with safety concerns and blocked sidewalks due to lack of regulation (Aizpuru et al., 2019; Allem & Majmundar, 2019; Gössling, 2020; James et al., 2019; Rahim Taleqani et al., 2019). Preliminary assessments indicate that e-scooters are about ten times more likely to get involved in an accident compared with bicycles (Fearnley et al., 2020). E-scooters are also linked

to environmental concerns due to the energy and materials used for their manufacturing combined with their short life cycle, but also due to energy and emissions arising from transporting the e-scooters to charging stations (Hollingsworth et al., 2019).

The impacts of e-scooters on travel behavior and the built environment are not clear yet since their use is very recent, and knowledge is still not mature enough to draw conclusions. Early studies indicate that the use of the e-scooter and its effects on travel behavior largely depend on the urban form, public transportation systems, and regulations on land use and transport. A recent in-depth study from Oslo, Norway, showed that e-scooters are mainly found in central, and usually denser, parts of the city and central intersections, and that they are typically used for "first-last mile" trips to work or education (Fearnley et al., 2020). The usage of escooters mainly replaced walking and to a smaller extent public transport, and more than half of the trips were made as part of a multimodal combination with other travel modes such as walking, metro, and bus (Fearnley et al., 2020). Studies from the United States report quite different results. A study from Washington, DC suggests that e-scooter sharing seems to be primarily used for recreation, leisure, and tourism activities and not for commuting, contrasting with bikesharing which is also used for commuting (McKenzie, 2019). Another study from Rosslyn, Virginia in the United States reported that the usage of e-scooters mainly replaced ridesourcing or taxi and walking, but also cycling, bus, and private car (James et al., 2019). This finding suggests that e-scooters can replace both motorized travel and active travel modes. Altogether, these early findings show that for compact urban forms, such as the central parts of Oslo, where the main travel modes are walking, cycling, and public transport (Mouratidis et al., 2019), the use of escooters mainly replaces these travel modes, while in more car-oriented contexts, e-scooters can replace higher percentages of car trips (ridesourcing, taxi, and private car), potentially resulting in lower car use, lower congestion, and less demand for parking. If e-scooters are used synergistically with public transport and active travel, they could also contribute to providing alternatives to car use and car ownership especially in car-oriented contexts. If used to promote sustainable mobility, by facilitating public transport and active travel and reducing car use, e-scooters could contribute to freeing up spaces occupied by cars and car infrastructure.

4.2.6. Mobility as a Service

Enabled by ICT – smartphones and mobile internet – Mobility as a Service (MaaS) offers a platform that connects and integrates different transport systems and suggests ideal transport options adjusted to the user's needs (Hietanen, 2014). Users of MaaS pay a single fee to use all the integrated transport options for a certain time period. MaaS aims to achieve an efficient transition from private car ownership to an integrated, multi-modal urban mobility (Hensher, 2017; Jittrapirom, Marchau et al., 2018; Li & Voege, 2017). The ideas of MaaS are not new. MaaS was preceded by ideas about connecting and integrating different systems and is now enabled by the advances in ICT (Lyons et al., 2019). MaaS is not a form of sharing economy per se, but takes advantage of the sharing economy, and shared mobility in particular, and aims to offer seamless, door-to-door, multi-modal mobility services via online platforms that bring together users and service operators (Hensher, 2017; Kamargianni et al., 2016; Li & Voege, 2017; Pangbourne et al., 2019).

Another concept that is related to MaaS is Mobility on Demand (MOD). MOD is a concept "where consumers can access mobility, goods, and services on demand by dispatching or using shared mobility, courier services, unmanned aerial vehicles (UAVs), and public transportation solutions. The most advanced forms of MOD passenger services incorporate trip planning and booking, real-time information, and fare payment into a single user interface." (Shaheen et al., 2017, p. 1). COVID-19 has promoted the on-demand delivery of goods and certain services, while it has slowed down the use of some shared mobility options.

Little knowledge exists so far on the effects of MaaS on travel behavior since it has not been applied yet on a large scale (Utriainen & Pöllänen, 2018). Cost efficiency and mode efficiency theoretically could lead to reductions in the use of the car. But MaaS results mainly depend on the goals that it aims to achieve and the accompanying policies and regulations (Moscholidou & Pangbourne, 2019; Sochor et al., 2018; Wong et al., 2019). Door-to-door, on-demand transport using some form of car travel (carsharing, ridesourcing, taxi) is neither in line with sustainable mobility nor with the development of environmentally friendly compact cities (Wong et al., 2019). If instead of public transport, people (are encouraged to) choose to use carsharing, ridesourcing or taxi then car use and traffic will increase (Hensher, 2017; Pangbourne et al., 2019). If on the other hand, MaaS implementation focuses on sustainable mobility, it could help decreasing the use of private cars and increasing the use of sustainable transport modes, according to a review of 31 relevant publications (Utriainen & Pöllänen, 2018). A MaaS implementation that focuses on sustainable mobility would be based on spatially efficient travel modes (Wong et al., 2019) and optimize the use of public space and the environmental and well-being benefits of sustainable mobility. If aimed towards sustainable mobility, MaaS can contribute to changes in the built environment including: reduction in cars, reduction in parking space, free space for other land uses (e.g. public space), and additional infrastructure for walking, cycling, and public transport. In this way, MaaS would be in line with compact city policies. If used in other ways, MaaS could favor decentralization and urban expansion by facilitating mobility in more remote areas.

5. Emerging transportation technologies

In addition to contributing to the dramatic increase in teleactivities and the sharing economy, ICT plays a crucial role in the development of emerging transportation technologies. For example, vehicle automation relies on ICT to function by using sensors and big data to analyze information and is also enabled by ICT for and during the vehicle usage (e.g. on-demand app services, apps for buying tickets, using smartphones, laptops, and tablets during trips). Emerging transportation technologies are either at a developing and testing stage or have recently been introduced to a limited extent but are expected to drastically change urban mobility in the future. Emerging transportation technologies include autonomous vehicles, air taxis, hoverbikes, drones, and robots (see e.g. European Commission, 2017). The use of these technologies is expected to influence travel behavior via substitution, complementarity, modification, and neutrality pathways. Moreover, depending on policy goals, on how technologies will be used, and on accompanying

Table 3

Summary of how emerging transportation technologies may influence travel behavior and the built environment.

Emerging transportation technologies	Travel behavior	Built environment	Source
Autonomous vehicles	 Private autonomous cars: can increase car travel, lead to more vehicle miles traveled, reduce public transport use, and reduce active travel Shared autonomous cars: seem to have similarities to carsharing, but with additional vehicle miles traveled Autonomous buses: could offer more frequent departures due to lower costs thus increasing the use of public transport and reducing car dependency 	 Private autonomous cars: increased congestion, potential urban expansion/urban sprawl, and reduced car parking Shared autonomous cars: may reduce parking space, but may also lead to parking relocation Autonomous buses might be in line with urbanization and compact city policies if their introduction is combined with appropriate policies 	Abe (2019); Ainsalu et al. (2018); Bösch et al. (2018); Carrese et al. (2019); Childress et al. (2015); Duarte and Ratti (2018); Fagnant and Kockelman (2014, 2015); Fraedrich et al. (2019); González-González et al. (2019); Kim et al. (2020a, 2020b); Litman (2020); Milakis et al. (2018); Milakis et al. (2017); Mouratidis and Cobeña Serrano (2021); Narayanan et al. (2020); Nazari et al. (2018); Nenseth et al. (2019); Nordhoff et al. (2018); Salonen and Haavisto (2019); Soteropoulos et al. (2019); Sperling (2018); Stead and Vaddadi (2019); Tirachini and Antoniou (2020); Zakharenko (2016); Zhang and Guhathakurta (2018); Zhang and Wang (2020)
Delivery drones	 Could reduce travel to stores and facilities But could have complementary rebound effects on travel by freeing up time for other activities 	 Would introduce air traffic into the aerial environment Could facilitate decentralization or urban expansion since they can increase accessibility to several material products 	Chiang et al. (2019); and authors' elaborations
Delivery robots	 Could replace some of the personal trips to stores and facilities But could have complementary rebound effects on travel by freeing up time for other activities 	• City infrastructure (e.g. sidewalks, pathways, building entrances, traffic regulations) may need to adapt to accommodate delivery robots	Authors' elaborations
Urban air mobility	 Increased mobility and accessibility Replacement of traditional transport modes Increased total travel distances due to quick access to destinations 	 Would introduce air traffic into the aerial environment Could facilitate decentralization or urban expansion as it has the potential to cover distances very quickly 	Al Haddad et al. (2020); Cohen et al. (2020); Fu et al. (2019); NASA (2018); Shaheen et al. (2018); and authors' elaborations

regulatory frameworks, emerging transportation technologies may influence transport systems, land uses, and location choices.

The following subsections review some of the major app-enabled emerging transportation technologies – autonomous vehicles, delivery drones, delivery robots, and urban air mobility – and how these relate to travel behavior and the built environment (see also Table 3 for an overview).

5.1. Autonomous vehicles

An autonomous vehicle, also known as automated vehicle, connected and autonomous vehicle, self-driving vehicle, or driverless vehicle, is a vehicle that is capable of sensing its environment and moving safely with little or no human interference (Taeihagh & Lim, 2019). Autonomous vehicles use artificial intelligence, sensors, and big data to analyze information, adapt to external conditions, and make decisions (Long et al., 2007; Taeihagh & Lim, 2019). Vehicle automation is classified into different categories based on its features ranging from 0 to 5; at level 5 the vehicle is expected to drive itself under all environmental conditions.

Autonomous vehicles can be private or shared (Fraedrich et al., 2019). Shared autonomous vehicles can be part of on-demand, shared mobility. Private autonomous cars can substitute conventional private cars, while shared autonomous cars can be used as taxis and rental cars or for carsharing purposes (Nazari et al., 2018). Autonomous vehicles can also be used as public transport modes in the form of autonomous buses (including minibuses and shuttles) (Ainsalu et al., 2018; Mouratidis & Cobeña Serrano, 2021; Salonen & Haavisto, 2019). Automated driving technology can be combined with electric vehicle technology, so autonomous vehicles can be electric in the form of private cars, shared cars, and public transport vehicles. Thus, overall, autonomous vehicles can be classified as (1) private autonomous cars, (2) shared autonomous cars/taxis, and (3) and autonomous buses (Bösch et al., 2018; Litman, 2020).

Vehicle automation will change the experience of private-car traveling since users will be able to engage in activities other than driving while traveling. Such activities may include eating or reading a book but also teleactivities such as teleworking, teleconferencing, or online entertainment (Burns, 2013). Technological development of autonomous vehicles has been rapid, but in order to

become a significant part of cities' daily mobility, further development is needed in terms of technology as well as governance and regulations to handle risks associated with safety, privacy, employment, environmental impacts, cybersecurity, and liability (Taeihagh & Lim, 2019).

Since full automation is still in a developing and testing phase, adequate data from the use of this technology are not yet available. Most studies are based on theoretical approaches, hypothetical models, and simulations. Based on these studies, autonomous cars are expected to increase vehicle miles traveled due to longer trips, shifts from public transport and active travel to the car, increased mobility especially for people with mobility restrictions, and more vehicle trips (Childress et al., 2015; Fagnant & Kockelman, 2014, 2015; Milakis et al., 2017). Increases in vehicle miles traveled are expected by both private and shared use of autonomous driving (Narayanan et al., 2020; Zhang & Guhathakurta, 2018). Policies accompanying autonomous vehicles are being considered such as per mile usage charges may counteract the increased travel distances (Childress et al., 2015). Parking relocation of shared autonomous vehicles may also induce environmental issues by generating a considerable amount of empty vehicle miles traveled (Zhang & Wang, 2020). Shared autonomous vehicles in the form of carsharing/taxi can offer the benefits of carsharing and potentially reduce private car usage and car ownership (Fagnant & Kockelman, 2014), but only if they compete with the private car and not with public transport.

Automated driving could be used to support public transport and not private mobility and car travel in the form of autonomous buses (Ainsalu et al., 2018; Fraedrich et al., 2019; Mouratidis & Cobeña Serrano, 2021; Nordhoff et al., 2018). Autonomous buses could have lower operational costs than conventional ones (Abe, 2019; Gkartzonikas & Gkritza, 2019; Tirachini & Antoniou, 2020); therefore they could be used to complement or substitute conventional public transport and could potentially increase the frequency of the departures (Bösch et al., 2018; Mouratidis & Cobeña Serrano, 2021). This scenario could result in an increased modal share of public transport and reduced car travel making this autonomous vehicle technology potentially positive for sustainable mobility (Litman, 2020; Nenseth, Ciccone, & Kristensen, 2019).

Autonomous vehicles are expected to not only change the way people travel, but also change the built environment in cities. Some studies have come up with different scenarios on how urban development will integrate autonomous vehicles and how the built environment will change because of this. Some scenarios suggest that autonomous vehicles may regenerate urban cores, reduce car parking, and free up spaces for other land uses, while other scenarios suggest that, by increasing mobility by car, autonomous vehicles will increase car dependency and lead to further urban sprawl (Sperling, 2018; Stead & Vaddadi, 2019). The implications of autonomous vehicles for urban development depend on how the technology will be used and the relevant policies. In other words, it will depend on whether automated driving technology will be used for private cars, for shared cars, or for public transport such as autonomous buses, but also on whether and how this use will be regulated and based on what goals it will be regulated (Soteropoulos et al., 2019; Stead & Vaddadi, 2019).

The adoption of autonomous vehicles may lead to higher efficiency in car parking and reduced space for car parking relieving land for other uses (Fagnant & Kockelman, 2015; Milakis et al., 2017; Zakharenko, 2016). Especially shared autonomous vehicles can reduce parking space, but also lead to parking relocation (Zhang & Wang, 2020). Private autonomous cars may increase accessibility for those who are not able or willing to drive, increase car use, and result in reduced use of public transport and active travel modes thus favoring residential relocation and urban sprawl (Carrese et al., 2019; Meyer et al., 2017; Soteropoulos et al., 2019; Zakharenko, 2016). According to international experts on accessibility issues, autonomous cars are expected to have two contrasting effects on urban form: densification of city centers and further urban sprawl (Milakis et al., 2018). The idea of autonomous cars leading to more urban sprawl has been questioned, as cities might attract even more residents if autonomous vehicles combined with sharing schemes manage to reduce congestion, pollution, and space used for car parking (Duarte & Ratti, 2018). Regarding shared autonomous vehicles in specific, it has been argued that although they will provide more freedom in choosing a residential location due to increased accessibility, this will not necessarily lead to urban sprawl because of commuting costs (Zhang & Guhathakurta, 2018).

Research has also examined the potential impacts of an era of autonomous vehicles on land use and travel by asking residents themselves. The majority of respondents in Georgia, USA expected that autonomous vehicles would not make them relocate or change the number of vehicles in their household, while some respondents imagine relocating or change vehicle ownership and travel behavior, depending on individual and household characteristics, neighborhood characteristics, and attitudes (Kim, Mokhtarian, & Circella, 2020a, 2020b).

To avoid urban expansion due to a potential introduction of autonomous vehicles, relevant urban and transport policies might be necessary. For environmentally friendly and livable urban development, scenario studies point to the use of autonomous vehicles combined with "mixed-use development policy, the clustering of urban facilities and services, the restriction of motorized access in cities and the adoption of shared high-quality multimodal transport" (González-González et al., 2019). If autonomous vehicle technology is used for public transport instead of car travel, urban expansion could be, at least partially, countered (Soteropoulos et al., 2019).

5.2. Delivery drones

Drones are driverless small aircraft that are also called unmanned aerial vehicles. They can function either using a remote control or autonomously like autonomous vehicles. The use of drones has been rapidly increasing. Current uses include aerial photography and video capturing; scientific research; police, surveillance, and military purposes; inspections of infrastructure; and recreation (Berkowitz, 2014). Several ethical, practical, and privacy issues are associated with drones (Lidynia et al., 2017; Murray & Chu, 2015). A new application of drones that is currently under development is the delivery of goods (Grippa et al., 2019; Pinto et al., 2019; Ulmer & Thomas, 2018). Drones are being developed to deliver products such as groceries, food from restaurants, mail, packages, medicines, vaccines, and equipment.

Using delivery drones in cities could have implications for urban mobility and travel behavior. Such uses of drones may provide easier access to certain products which could reduce travel to stores and facilities. This could result in some reductions in transport emissions and costs (Chiang et al., 2019). As in the case of certain teleactivities, time saved from traveling to such stores could be spent on travel for other purposes. Yet, there are not enough insights into how exactly drones could influence travel behavior. Potential impacts also largely depend on accompanying policies and regulations.

Drones can cause drastic changes in the urban built environment by becoming part of the aerial environment in cities. Increased areal transport may also lead to a decrease in land transport, freeing up space for alternative land uses. Such a form of transport may also create new urban soundscapes, with possible negative impacts. The exact impacts of all these changes are hard to be foreseen. If drones are to fly within or between cities to deliver goods, strong policies and regulations would be needed to reduce associated risks. Delivery drones could potentially facilitate decentralization or urban expansion since they can increase accessibility to several material products. The implications of delivery drones for the built environment have not been sufficiently examined by existing research.

5.3. Delivery robots

In addition to delivering goods with drones, mobile robots are also being developed for this purpose (Buchegger et al., 2018; Raza et al., 2019). Delivery robots take the form of small vehicles or legged walking robots. They can deliver several types of material products such as food, mail, packages, and medicines. Delivery robots are so far designed for last-mile delivery and take-out services. Several startup companies have already developed such robots and pilot projects are already applied in various cities (e.g. San Francisco, Berkeley, and Los Angeles in the United States and Milton Keynes in the United Kingdom).

Similarly to delivery drones, delivery robots may increase accessibility to goods, especially for people with mobility difficulties, people who are not in proximity to certain stores and facilities or people with time restrictions. As with delivery drones, delivery robots may replace some of the personal trips to stores and facilities but could also have complementary rebound effects on travel by freeing up time for other activities. City infrastructure would need to adapt to accommodate delivery robots. For example, sidewalks and pathways would need to enable the unobstructed movement of the robots, building entrances might have to be adjusted, and traffic regulations may also be necessary. These adjustments would depend on the level of automation of the robots. Highly developed robots may not need heavily standardized environments. Since, for the moment at least, delivery robots are not expected to travel long distances, location choices and spatial structure are not expected to be significantly influenced by delivery robots. However, the impacts of delivery robots on both travel behavior and the built environment could rapidly grow and should be further studied.

5.4. Urban air mobility

Urban air mobility is the concept of moving people by air within cities (Al Haddad et al., 2020; Fu et al., 2019). Transport modes of urban air mobility are in a development and testing phase, and progress is rapid. Urban air mobility can be private or shared. It can also be on-demand and automated. Examples of urban air mobility are personal air vehicles, air taxis, and hoverbikes. There is a growing body of literature on urban air mobility that focuses on challenges and barriers such as societal acceptance, safety, equity, weather, legal/regulatory, among others (e.g. Cohen et al., 2020; Serrao et al., 2018; Shaheen et al., 2018). Although the familiarity with the urban air mobility concept is still relatively low among the public, there is an important minority of potential consumers in the United States that would consider traveling by urban air mobility options (NASA, 2018; Shaheen et al., 2018). Public concerns about urban air mobility are generally related to safety, privacy, unemployment, environment, noise, and pollution (NASA, 2018).

Urban air mobility has the potential to cause unpreceded changes in urban mobility as we know it today. When relevant technology develops, it could allow people to cover long distances very quickly (Cohen et al., 2020). Therefore, the distance traveled could dramatically increase. Increased accessibility offered by urban air mobility could result in replacing part of the usage of traditional travel modes. At the same time, covering long distances so easily could facilitate living in more remote places or on the outskirts of cities, and this could encourage sprawled development. Urban air mobility would cause changes in the aerial environment in cities and regions as it would introduce air traffic. Peer-reviewed research examining potential impacts of urban air mobility on travel behavior and the built environment is scarce (Cohen et al., 2020), but urgently needed, since technology is developing rapidly.

6. Conclusions

6.1. Summary of findings and discussion

In this paper, we synthesized the state of knowledge on the implications of teleactivities, the sharing economy, and emerging transportation technologies for travel behavior and the built environment. Teleactivities, the sharing economy, and emerging transportation technologies have three common characteristics. They are enabled by the use of online apps by residents and visitors, are interconnected, and may influence both the built environment and travel behavior. As an umbrella term for the emerging trends found in literature, we may call cities encompassing these three app-enabled elements "App Cities". We believe that such a concept could be helpful in highlighting the interrelatedness of these trends and their accumulative impact on the built environment and travel behavior.

The synthesis of literature presented in the paper shows that App Cities already have important implications for travel behavior by altering modal shares, travel distances, trip purposes, and trip frequencies. Further changes are expected with the rapidly increasing adoption of teleactivities and engagement in the sharing economy, while emerging transportation technologies will cause drastic

changes in urban mobility as we know it today. At the same time, the paper demonstrates that teleactivities, sharing economy, and emerging smart mobility options are changing the built environment in cities: (a) by becoming part of it as new transport systems, (b) the built environment is being transformed to accommodate them (e.g. new city infrastructure), (c) the changes in people's lifestyles and behaviors lead to subsequent changes in the built environment including changes in location choices and land uses.

Our findings addressing the first research question of the paper – the implications of teleactivities, sharing economy, and emerging transportation technologies for travel behavior – can be summarized as follows. Telecommuting and teleconferencing may result in reductions in total travel distances. Online shopping, online education, teleleisure, telehealth, and online social networking do not seem to reduce overall travel distances, as time saved from physical travel for these activities may free up time for traveling to reach other activities. Bikesharing seems to increase cycling modal share, replace the use of public transport, walking, and car, and increase total active travel; and thereby it seems to reduce vehicle miles traveled. E-bike sharing may further replace the use of public transport and car and could in turn further reduce vehicle miles traveled. Carsharing, if not intended to replace regular use of private cars with regular use of shared cars, could be conducive to increases in walking, cycling, and use of public transport and reductions in private car use, private car ownership, and vehicle miles traveled. Ridesharing in the form of carpooling or vanpooling may increase mobility and car travel efficiency but is competitive to and may replace public transport. Ridesourcing, as an informal, cheaper form of taxi, could give rise to more car travel. E-scooters and other forms of micro-mobility can facilitate the use of public transport as a first-last mile option, while they replace active travel, public transport, and car travel. Accommodation sharing may increase total travel in three ways: by displacing residents to less central locations, by inducing additional long-distance travel for tourism and business, and by providing hosts with extra income that could, in turn, incentivize them to engage in more frequent long-distance trips or holidays in destinations that are further away. Private autonomous cars can increase car travel, induce extra vehicle miles traveled, and reduce public transport usage and active travel. Shared autonomous vehicles seem to have implications similar to carsharing, but with additional vehicle miles traveled. Autonomous buses could offer more frequent departures due to lower costs thus having the potential to trigger further use of public transport and to reduce car dependency. MaaS is largely dependent on its goals. If steered towards sustainable mobility, it could provide incentives for using active travel and public transport, while discouraging car travel, and then it could contribute to reductions in car use and car ownership. If steered only towards greater car mobility, cost reductions, and comfort by offering door-to-door, on-demand transport with some form of car travel (carsharing, ridesourcing, taxi), then it could contribute to increased car travel and vehicle miles traveled. Delivery drones and delivery robots would facilitate access to goods and could reduce travel to shops, but they might have rebound effects on other forms of personal travel. Urban air mobility with personal air vehicles, air taxis, and hoverbikes may cause dramatic changes in travel behavior in cities. Potential impacts include increased mobility and accessibility, replacement of traditional transport modes, and increased total travel distances.

Our findings addressing the second research question of the paper - the implications of teleactivities, sharing economy, and emerging transportation technologies for the built environment - are classified according to three pathways. (a) New transport modes including shared bikes, shared cars, and shared e-scooters are now integrated into the built environment, while emerging transport modes will also be integrated into the built environment: autonomous cars and autonomous buses, delivery drones, deliver robots, personal air vehicles, air taxis, and hoverbikes. (b) The built environment is changing to accommodate the elements of the App City. New infrastructure is or will be developed to accommodate bikesharing and e-scooter sharing systems, or urban air mobility. Sidewalks and pathways could be adjusted to facilitate the movement of delivery robots. Roads could be adapted to facilitate the movement of autonomous vehicles. (c) Behavioral changes due to teleactivities, the sharing economy, and emerging transportation technologies may have a subsequent influence on the built environment. Modal share changes could lead to changes in land use and transport infrastructure. Reductions in car use and ownership could result in more efficient use of space by freeing up space due to reduced needs for road capacity and car parking. Free space could result in new land uses. On the other hand, increased car use could result in more congestion and expansion of car infrastructure. Online access to certain goods and services is conducive to the decline or disappearance of certain "brick and mortar" shops and facilities. Increased long-distance travel, partially induced by accommodation sharing platforms such as Airbnb, can contribute to more tourists and business travelers and higher demands for city infrastructure and building capacity. Increased accessibility and mobility due to ICT, sharing systems, and new transport modes could enable remote residential locations or workplace locations, thereby triggering decentralization and urban expansion, if spatial development is not regulated.

Based on the findings from our literature review we can draw some preliminary conclusions on the potential of the App City components to contribute to sustainable mobility. Among the strategies for achieving sustainable mobility are the shift towards more environmentally friendly travel modes and the reduction of travel distances with appropriate land use development (see e.g. Banister, 2008). The implications of the App City components for sustainable mobility (in terms of travel mode and total travel distances) could be classified as: positive for bikesharing; potentially positive for autonomous buses, carsharing, telecommuting, and teleconferencing; potentially negative for private autonomous cars, ridesourcing such as Uber, accommodation sharing such as Airbnb, and urban air mobility; mixed (due to possible rebound effects) for online shopping, online education, teleleisure, telehealth, and online social networking; mixed/context-dependent for e-scooter sharing and ridesharing; largely dependent on goals, use, and regulations for Mobility as a Service; and relatively unknown (but possibly mixed due to rebound effects) for delivery drones and delivery robots. It should be noted that even potentially eco-friendly technologies and mobility services should be accompanied by strong policies and regulatory frameworks in order to promote sustainable mobility (e.g. to reduce car driving and prevent urban sprawl) (Lyons et al., 2018; Moscholidou & Pangbourne, 2019; Stead & Vaddadi, 2019; Wong et al., 2019).

App City developments might occur more quickly than expected due to the COVID-19 pandemic in 2019–2020. This global crisis has made even more evident the potential of certain elements of App Cities, especially teleactivities but also to some extent transportation technologies such as drones and robots, to reduce the negative impacts of the crisis. During the "lockdowns" that occurred in several geographical areas, people who had access to the internet were enabled or forced to virtually perform, to the extent possible, a

wide range of activities including accessing certain health services, paying bills, following classes of physical exercise, socializing with friends and relatives, engaging in several forms of online recreation, and even working or studying when possible using teleworking and online education options. In addition, certain emerging transportation technologies were relevant during the COVID-19 crisis. In several cities worldwide, drones and robots were used for delivering products such as food, paper, and medicines; sanitizing and disinfecting; and monitoring the restrictive measures during the lockdowns. Although COVID-19 has boosted the adoption of teleactivities, and to some extent drones and robots, it may have slowed down the development and use of some shared mobility options and may have unclear effects on other emerging transportation technologies and automation.

6.2. Future research directions

Due to the rapidly evolving changes in ICT and transport technology, researchers are being seriously challenged to produce constantly updated knowledge. Emerging from this paper, we list the most eminent research directions about the implications of App Cities for travel behavior and the built environment. We grouped them into two categories: substantive and methodological.

6.2.1. Substantive research directions

There is a need to study the constantly transforming built environment in App Cities (see conceptual model in Section 2, the link between App City and Built environment) and especially how exactly the behavioral changes due to different App City components are influencing the different built environment factors and their interplay (see conceptual model in Section 2, the link between Travel behavior and Built environment). The lack of a more nuanced understanding of these links surfaced for all three App City components discussed in this paper. Understanding these influences in more nuanced ways is essential in order to be able to steer and coordinate these developments, mitigate negative impacts, and support App City components that are in line with policy goals.

The implications of certain teleactivities for travel behavior and the built environment need to be studied in more depth: online education, teleconferencing, teleleisure, telehealth, and online social networking (see Section 3, Table 1). Existing knowledge on these implications is ambiguous, in particular because of the under-researched problematic effects of teleactivities such rebound effects or possible effects on urban sprawl. A comprehensive and nuanced understanding of both the positive and the negative impacts of teleactivities on sustainable mobility is necessary in order to inform planning policy and decision making within this area.

More in-depth studies are needed on the evolving shared mobility options including carsharing, ridesourcing, and e-scooter sharing (see Section 4, Table 2). Also here, existing research shows ambiguity about possible effects, and results seem to be highly context-dependent. In particular, the positive and the negative impacts of these options, in different contexts and in relation to sustainable mobility goals, need to be disentangled and assessed.

Studies should place more emphasis on ongoing pilot projects of emerging transportation technologies: autonomous cars and buses, Mobility as a Service, delivery robots, and urban air mobility (see Section 5). Further, there seems to be a general lack of research examining emerging transportation technologies such as urban air mobility, delivery robots, and delivery drones. Scenario studies could be a powerful tool to critically assess their implications (see Section 5, Table 3).

Across all sections, the paper has shown that the impacts of App City components are strongly dependent on policy and governance tools to steer urban development. Different existing government models should be compared regarding their influence on different possible pathways for App City developments.

Another research need is to continuously investigate how App City developments relate to safety, social equity, gender issues, wellbeing, and environmental outcomes. These implications of App Cities have not been investigated in-depth in the paper, but they are clearly important avenues for future research that need critical exploration. Future research could provide a synthesis of the magnitude of each technology on urban form and environmental and social sustainability and discuss the outcomes of possible future scenarios.

Although the paper has explained the interplay between App City components and has examined how this might influence travel behavior and the built environment (Sections 2–5), the main aim has been to present an overview of each component individually. Future studies could examine the implications of the interplay between App City components in greater depth.

Finally, App City developments should be studied in relation to disruptions in travel and activity behavior, such as the COVID-19 pandemic outbreak (Section 6.1). Future research could explore how App Cities changed during and after this crisis and how travel behavior and built environments have been affected and will be affected. We expect disruptions like pandemics to affect society way less, thanks to options provided by the App City, at least options for teleactivities. The social and environmental implications of changes in App Cities due to the pandemic crisis also need to be explored.

6.2.2. Methodological research directions

There is a need for more complex research designs including before-after studies, panel studies, and (quasi) experimental studies to provide a more robust understanding of changes in travel behavior due to teleactivities, sharing economy, and emerging transportation technologies (see for example Section 3.1).

There is a need for more complex statistical models of travel behavior that would include geo-referenced data on and factors like attitudes, and personality and that would be analyzed with more sophisticated statistical techniques such as structural equation modeling.

The large majority of the studies reviewed in this paper reports findings that are based on quantitative research. These studies are helpful to identify possible impacts and assess the strength of the relationships identified in our conceptual model. There is, however, a strong need for qualitative research to understand causal pathways. Qualitative or mixed-methods studies are necessary in order to not only identify possible impacts, but to disentangle and explain the causal mechanism behind patterns found in quantitative analyses.

K. Mouratidis et al.

Finally, and closely linked to the previous research need, the relationship between App City developments, travel behavior, and changes in the built environment may be strongly context-dependent. There is a need for research to be context-sensitive, for example conducting comparative studies in diverse cultural, social, and governmental settings. This would also help to understand the partly contradictory findings of some existing studies.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

The study is part of the research project "App Cities: New urban technologies, daily travel, and quality of life", funded by the Norwegian University of Life Sciences. We thank the three anonymous reviewers for their insightful comments. We are also grateful to Editors Robert B. Noland and Xinyu (Jason) Cao for their valuable suggestions on earlier versions of the paper.

Appendix

Literature search

The paper reviews around 200 studies. To keep the literature scope more manageable, we focus mostly on academic peer-reviewed articles, written in English, but we also include other sources like books, book chapters, reports, and conference papers, especially in the absence of peer-reviewed literature. Our literature search commenced by identifying recent literature review papers, when available, and was supplemented with a systematic search in Scopus. The next step in our literature search was backward snowballing (van Wee & Banister, 2016). To perform backward snowballing, we departed from relevant review papers for topics that have been sufficiently explored by previous research. For topics that were underexplored, we used the few relevant papers found in our Scopus searches as points of departure for backward snowballing. The search string comprised keywords for the targeted App City element in combination with keywords for the implications investigated in the paper. For example, in our Scopus search for e-scooter sharing, we used the following search string (("scooter sharing" OR "scooter-sharing" OR "scooter share" OR "location")). For topics that are covered by large amounts of literature such as autonomous vehicles or telecommuting, we reviewed relevant literature review papers, seminal papers, and highly relevant papers to keep the size of the literature manageable. Several related papers have naturally been omitted during this process. For topics that are not covered by literature or when literature is scarce (e.g. emerging mobility technologies like delivery drones, delivery robots, or urban air mobility), we discuss the research questions of the study mostly at a conceptual level and recommend directions for future work.

References

Aguiléra, A., Guillot, C., Rallet, A., 2012. Mobile ICTs and physical mobility: review and research agenda. Transport. Res. Part A: Policy Pract. 46 (4), 664–672. https://doi.org/10.1016/j.tra.2012.01.005.

Ainsalu, J., Arffman, V., Bellone, M., Ellner, M., Haapamäki, T., Haavisto, N., Åman, M., 2018. State of the art of automated buses. Sustainability 10 (9). https://doi.org/10.3390/su10093118.

Aizpuru, M., Farley, K.X., Rojas, J.C., Crawford, R.S., Moore Jr., T.J., Wagner, E.R., 2019. Motorized scooter injuries in the era of scooter-shares: a review of the national electronic surveillance system. Am. J. Emerg. Med. 37 (6), 1133–1138. https://doi.org/10.1016/j.ajem.2019.03.049.

Al Haddad, C., Chaniotakis, E., Straubinger, A., Plötner, K., Antoniou, C., 2020. Factors affecting the adoption and use of urban air mobility. Transport. Res. Part A: Policy Pract. 132, 696–712. https://doi.org/10.1016/j.tra.2019.12.020.

Alaimo, L.S., Fiore, M., Galati, A., 2020. How the Covid-19 pandemic is changing online food shopping human behaviour in Italy. Sustainability 12 (22), 9594. Alemi, F., Circella, G., Handy, S., Mokhtarian, P., 2018. What influences travelers to use Uber? Exploring the factors affecting the adoption of on-demand ride services

in California. Travel Behav. Soc. 13, 88–104. https://doi.org/10.1016/j.tbs.2018.06.002. Allem, J.-P., Majmundar, A., 2019. Are electric scooters promoted on social media with safety in mind? A case study on Bird's Instagram. Prevent. Med. Rep. 13,

62–63. https://doi.org/10.1016/j.pmedr.2018.11.013.

Andreev, P., Salomon, I., Pliskin, N., 2010. Review: State of teleactivities. Transport. Res. Part C: Emerg. Technol. 18 (1), 3–20. https://doi.org/10.1016/j. trc.2009.04.017.

Balampanidis, D., Maloutas, T., Papatzani, E., Pettas, D., 2019. Informal urban regeneration as a way out of the crisis? Airbnb in Athens and its effects on space and society. Urban Res. Pract. 1–20 https://doi.org/10.1080/17535069.2019.1600009.

Banister, D., 2008. The sustainable mobility paradigm. Transp. Policy 15 (2), 73–80. https://doi.org/10.1016/j.tranpol.2007.10.005.

Banister, D., Stead, D., 2004. Impact of information and communications technology on transport. Transp. Rev. 24 (5), 611–632. https://doi.org/10.1080/ 0144164042000206060.

Battarra, R., Gargiulo, C., Tremiterra, M.R., Zucaro, F., 2018. Smart mobility in Italian metropolitan cities: a comparative analysis through indicators and actions. Sustain. Cities Soc. 41, 556–567. https://doi.org/10.1016/j.scs.2018.06.006.

Batty, M., 2020. How disruptive are new urban technologies? Environ. Plan. B: Urban Anal. City Sci. 47 (1), 3-6. https://doi.org/10.1177/2399808320902574.

Abe, R., 2019. Introducing autonomous buses and taxis: Quantifying the potential benefits in Japanese transportation systems. Transport. Res. Part A: Policy Pract. 126, 94–113. https://doi.org/10.1016/j.tra.2019.06.003.

Berkowitz, R., 2014. Drones and the question of "The Human". Ethics Int. Affairs 28 (2), 159–169. https://doi.org/10.1017/S0892679414000185.

Botsman, R., Rogers, R., 2010. What's Mine is Yours: The Rise of Collaborative Consumption. HarperCollins Publishers, New York, NY.

Brown, A.E., 2020. Who and where rideshares? Rideshare travel and use in Los Angeles. Transport. Res. Part A: Policy Pract. 136, 120–134. https://doi.org/10.1016/j.tra.2020.04.001.

- Buchegger, A., Lassnig, K., Loigge, S., Mühlbacher, C., Steinbauer, G., 2018. An autonomous vehicle for parcel delivery in urban areas. Paper Presented at the 21st International Conference on Intelligent Transportation Systems (ITSC).
- Bulu, S.T., 2012. Place presence, social presence, co-presence, and satisfaction in virtual worlds. Comput. Educ. 58 (1), 154–161. https://doi.org/10.1016/j. compedu.2011.08.024.

Burns, L.D., 2013. A vision of our transport future. Nature 497 (7448), 181-182. https://doi.org/10.1038/497181a.

- Bösch, P.M., Becker, F., Becker, H., Axhausen, K.W., 2018. Cost-based analysis of autonomous mobility services. Transp. Policy 64, 76–91. https://doi.org/10.1016/j. tranpol.2017.09.005.
- Cao, X., Chen, Q., Choo, S., 2013. Geographic distribution of E-shopping: application of structural equation models in the twin cities of minnesota. Transp. Res. Rec. 2383 (1), 18–26. https://doi.org/10.3141/2383-03.
- Cao, X., Douma, F., Cleaveland, F., 2010. Influence of E-shopping on shopping travel: evidence from minnesota's twin cities. Transp. Res. Rec. 2157 (1), 147–154. https://doi.org/10.3141/2157-18.
- Cao, X., Xu, Z., Douma, F., 2012. The interactions between e-shopping and traditional in-store shopping: an application of structural equations model. Transportation 39 (5), 957–974. https://doi.org/10.1007/s11116-011-9376-3.
- Carrese, S., Nigro, M., Patella, S.M., Toniolo, E., 2019. A preliminary study of the potential impact of autonomous vehicles on residential location in Rome. Res. Transport. Econ. 75, 55–61. https://doi.org/10.1016/j.retrec.2019.02.005.
- Castro, A., Gaupp-Berghausen, M., Dons, E., Standaert, A., Laeremans, M., Clark, A., Götschi, T., 2019. Physical activity of electric bicycle users compared to conventional bicycle users and non-cyclists: insights based on health and transport data from an online survey in seven European cities. Transport. Res. Interdiscip. Perspect. 1, 100017. https://doi.org/10.1016/j.trip.2019.100017.

Chan, N.D., Shaheen, S.A., 2012. Ridesharing in north america: past, present, and future. Transp. Rev. 32 (1), 93-112. https://doi.org/10.1080/ 01441647.2011.621557.

Chiang, W.-C., Li, Y., Shang, J., Urban, T.L., 2019. Impact of drone delivery on sustainability and cost: realizing the UAV potential through vehicle routing optimization. Appl. Energy 242, 1164–1175. https://doi.org/10.1016/j.apenergy.2019.03.117.

Childress, S., Nichols, B., Charlton, B., Coe, S., 2015. Using an activity-based model to explore the potential impacts of automated vehicles. Transp. Res. Rec. 2493 (1), 99–106. https://doi.org/10.3141/2493-11.

Choo, S., Mokhtarian, P.L., Salomon, I., 2005. Does telecommuting reduce vehicle-miles traveled? An aggregate time series analysis for the U.S. Transportation 32 (1), 37–64. https://doi.org/10.1007/s11116-004-3046-7.

Circella, G., Mokhtarian, P.L., 2017. Impacts of information and communication technology. Geograp. Urban Transport. 86-111.

Cohen, A., Guan, J., Beamer, M., Dittoe, R., Mokhtarimousavi, S., 2020. Reimagining the Future of Transportation with Personal Flight: Preparing and Planning for Urban Air Mobility. Transportation Sustainability Research Center, UC Berkeley.

- de Graaff, T., Rietveld, P., 2007. Substitution between working at home and out-of-home: the role of ICT and commuting costs. Transport. Res. Part A: Policy Pract. 41 (2), 142–160. https://doi.org/10.1016/j.tra.2006.02.005.
- De Vos, D., Van Ham, M., & Meijers, E.J., 2019. Working from Home and Commuting: Heterogeneity over Time, Space, and Occupations. IZA Discussion Paper No. 12578.
- Denstadli, J.M., 2004. Impacts of videoconferencing on business travel: the Norwegian experience. J. Air Transp. Manage. 10 (6), 371–376. https://doi.org/10.1016/ j.jairtraman.2004.06.003.
- Di Marino, M., Lilius, J., Lapintie, K., 2018. New forms of multi-local working: identifying multi-locality in planning as well as public and private organizations' strategies in the Helsinki region. Eur. Plan. Stud. 26 (10), 2015–2035. https://doi.org/10.1080/09654313.2018.1504896.

Ding, Y., Lu, H., 2017. The interactions between online shopping and personal activity travel behavior: an analysis with a GPS-based activity travel diary.

Transportation 44 (2), 311–324. https://doi.org/10.1007/s11116-015-9639-5.
Dogru, T., Hanks, L., Mody, M., Suess, C., Sirakaya-Turk, E., 2020. The effects of Airbnb on hotel performance: evidence from cities beyond the United States. Tourism Manage. 79, 104090. https://doi.org/10.1016/j.tourman.2020.104090.

Dorsey, E.R., Topol, E.J., 2016. State of telehealth. N. Engl. J. Med. 375 (2), 154-161. https://doi.org/10.1056/NEJMra1601705.

Duarte, F., Ratti, C., 2018. The impact of autonomous vehicles on cities: a review. J. Urban Technol. 25 (4), 3–18. https://doi.org/10.1080/10630732.2018.1493883.
Duran-Rodas, D., Chaniotakis, E., Antoniou, C., 2019. Built environment factors affecting bike sharing ridership: data-driven approach for multiple cities. Transp. Res.
Rec. 0361198119849908 https://doi.org/10.1177/0361198119849908.

Eckhardt, G.M., Bardhi, F., 2015. The sharing economy isn't about sharing at all. Harvard Bus. Rev. 28 (1), 2015.

Erhardt, G.D., Roy, S., Cooper, D., Sana, B., Chen, M., Castiglione, J., 2019. Do transportation network companies decrease or increase congestion? Sci. Adv. 5 (5), eaau2670. https://doi.org/10.1126/sciadv.aau2670.

Ettema, D., 2018. Apps, activities and travel: an conceptual exploration based on activity theory. Transportation 45 (2), 273–290. https://doi.org/10.1007/s11116-017-9844-5.

- European Commission, 2017. Smart mobility and services. Retrieved from Luxemburg: https://ec.europa.eu/transparency/regexpert/index.cfm?do=groupDetail.groupDetailDoc&id=34596&no=1.
- Fagnant, D.J., Kockelman, K., 2014. The travel and environmental implications of shared autonomous vehicles, using agent-based model scenarios. Transport. Res. Part C: Emerg. Technol. 40, 1–13. https://doi.org/10.1016/j.trc.2013.12.001.
- Fagnant, D.J., Kockelman, K., 2015. Preparing a nation for autonomous vehicles: opportunities, barriers and policy recommendations. Transport. Res. Part A: Policy Pract. 77, 167–181. https://doi.org/10.1016/j.tra.2015.04.003.
- Farag, S., Krizek, K.J., Dijst, M., 2006. E-shopping and its relationship with in-store shopping: empirical evidence from the Netherlands and the USA. Transp. Rev. 26 (1), 43–61. https://doi.org/10.1080/01441640500158496.

Fearnley, N., Berge, S.H., & Johnsson, E., 2020. Delte elsparkesykler i Oslo. Retrieved from Oslo: https://www.toi.no/getfile.php?mmfileid=52254.

Ferrell, C.E., 2004. Home-based teleshoppers and shopping travel: do teleshoppers travel less? Transp. Res. Rec. 1894 (1), 241-248. https://doi.org/10.3141/1894-25.

Ferrero, F., Perboli, G., Rosano, M., Vesco, A., 2018. Car-sharing services: an annotated review. Sustain. Cities Soc. 37, 501–518. https://doi.org/10.1016/j. scs.2017.09.020.

Fishman, E., 2016. Bikeshare: a review of recent literature. Transp. Rev. 36 (1), 92–113. https://doi.org/10.1080/01441647.2015.1033036.

Fishman, E., Cherry, C., 2016. E-bikes in the mainstream: reviewing a decade of research. Transp. Rev. 36 (1), 72–91. https://doi.org/10.1080/

01441647.2015.1069907.

Fishman, E., Washington, S., Haworth, N., 2013. Bike share: a synthesis of the literature. Transp. Rev. 33 (2), 148–165. https://doi.org/10.1080/ 01441647.2013.775612.

- Fishman, E., Washington, S., Haworth, N., 2014. Bike share's impact on car use: evidence from the United States, Great Britain, and Australia. Transport. Res. Part D: Transp. Environ. 31, 13–20. https://doi.org/10.1016/j.trd.2014.05.013.
- Fishman, E., Washington, S., Haworth, N., 2015. Bikeshare's impact on active travel: evidence from the United States, Great Britain, and Australia. J. Transp. Health 2 (2), 135–142. https://doi.org/10.1016/j.jth.2015.03.004.
- Fraedrich, E., Heinrichs, D., Bahamonde-Birke, F.J., Cyganski, R., 2019. Autonomous driving, the built environment and policy implications. Transport. Res. Part A: Policy Pract. 122, 162–172. https://doi.org/10.1016/j.tra.2018.02.018.

- Freathy, P., Calderwood, E., 2013. The impact of internet adoption upon the shopping behaviour of island residents. J. Retail. Consumer Ser. 20 (1), 111–119. https://doi.org/10.1016/j.jretconser.2012.10.012.
- Fu, M., Rothfeld, R., Antoniou, C., 2019. Exploring preferences for transportation modes in an urban air mobility environment: munich case study. Transp. Res. Rec. 2673 (10), 427–442. https://doi.org/10.1177/0361198119843858.
- Fyhri, A., Fearnley, N., 2015. Effects of e-bikes on bicycle use and mode share. Transport. Res. Part D: Transp. Environ. 36, 45–52. https://doi.org/10.1016/j. trd.2015.02.005.
- Gajendran, R.S., Harrison, D.A., 2007. The good, the bad, and the unknown about telecommuting: meta-analysis of psychological mediators and individual consequences. J. Appl. Psychol. 92 (6), 1524–1541. https://doi.org/10.1037/0021-9010.92.6.1524.
- Gansky, L., 2010. The Mesh: Why the Future of Business is Sharing. Penguin, New York, NY
- Geitmann, A., 2020. Travel less. Make it worthwhile. Cell 182 (4), 790–793. https://doi.org/10.1016/j.cell.2020.07.026.
- Gkartzonikas, C., Gkritza, K., 2019. What have we learned? A review of stated preference and choice studies on autonomous vehicles. Transport. Res. Part C: Emerg. Technol. 98, 323–337. https://doi.org/10.1016/j.trc.2018.12.003.
- Golden, T.D., Veiga, J.F., Dino, R.N., 2008. The impact of professional isolation on teleworker job performance and turnover intentions: does time spent teleworking, interacting face-to-face, or having access to communication-enhancing technology matter? J. Appl. Psychol. 93 (6), 1412–1421. https://doi.org/10.1037/a0012722.
- González-González, E., Nogués, S., Stead, D., 2019. Automated vehicles and the city of tomorrow: a backcasting approach. Cities 94, 153–160. https://doi.org/ 10.1016/j.cities.2019.05.034.
- Grippa, P., Behrens, D.A., Wall, F., Bettstetter, C., 2019. Drone delivery systems: job assignment and dimensioning. Autonomous Robots 43 (2), 261–274. https://doi.org/10.1007/s10514-018-9768-8.
- Gubins, S., van Ommeren, J., de Graaff, T., 2019. Does new information technology change commuting behavior? Ann. Reg. Sci. 62 (1), 187–210. https://doi.org/ 10.1007/s00168-018-0893-2.
- Guerin, T.F., 2017. A demonstration of how virtual meetings can enhance sustainability in a corporate context. Environ. Qual. Manage. 27 (1), 75–81. https://doi.org/ 10.1002/tgem.21515.
- Guidon, S., Becker, H., Dediu, H., Axhausen, K.W., 2019. Electric Bicycle-sharing: a new competitor in the urban transportation market? An empirical analysis of transaction data. Transp. Res. Rec. 2673 (4), 15–26. https://doi.org/10.1177/0361198119836762.
- Gurran, N., Phibbs, P., 2017. When tourists move in: how should urban planners respond to airbnb? J. Am. Plan. Assoc. 83 (1), 80–92. https://doi.org/10.1080/01944363.2016.1249011.
- Guttentag, D., 2019. Progress on Airbnb: a literature review. J. Hospit. Tourism Technol. 10 (4), 814-844. https://doi.org/10.1108/JHTT-08-2018-0075.
- Gössling, S., 2018. ICT and transport behavior: a conceptual review. Int. J. Sustain. Transport. 12 (3), 153–164. https://doi.org/10.1080/15568318.2017.1338318.
 Gössling, S., 2020. Integrating e-scooters in urban transportation: Problems, policies, and the prospect of system change. Transport. Res. Part D: Transp. Environ. 79, 102230. https://doi.org/10.1016/j.trd.2020.102230.
- Hampshire, R.C., Gaites, C., 2011. Peer-to-peer carsharing: market analysis and potential growth. Transp. Res. Rec. 2217 (1), 119-126. https://doi.org/10.3141/ 2217-15.
- Harris, M., Gorenflo, N., 2012. Share or die: Voices of the Get Lost Generation in the Age of Crisis. New Society Publishers, Gabriola Island, BC.

Harun, M.H., 2001. Integrating e-Learning into the workplace. Int. Higher Educ. 4 (3), 301-310. https://doi.org/10.1016/S1096-7516(01)00073-2.

- He, S.Y., Hu, L., 2015. Telecommuting, income, and out-of-home activities. Travel Behav. Soc. 2 (3), 131-147. https://doi.org/10.1016/j.tbs.2014.12.003.
- Helminen, V., Ristimäki, M., 2007. Relationships between commuting distance, frequency and telework in Finland. J. Transp. Geogr. 15 (5), 331-342. https://doi.org/10.1016/j.jtrangeo.2006.12.004.
- Henao, A., Marshall, W.E., 2019a. The impact of ride-hailing on vehicle miles traveled. Transportation 46 (6), 2173–2194. https://doi.org/10.1007/s11116-018-9923-2.
- Henao, A., Marshall, W.E., 2019b. The impact of ride hailing on parking (and vice versa). J. Transp. Land Use 12 (1), 127–147. https://doi.org/10.2307/26911261. Hensher, D.A., 2017. Future bus transport contracts under a mobility as a service (MaaS) regime in the digital age: are they likely to change? Transport. Res. Part A: Policy Pract. 98, 86–96. https://doi.org/10.1016/j.tra.2017.02.006.
- Herring, H., Roy, R., 2002. Sustainable services, electronic education and the rebound effect. Environ. Impact Assess. Rev. 22 (5), 525–542. https://doi.org/10.1016/ S0195-9255(02)00026-4.
- Hietanen, S., 2014. Mobility as a service the new transport model? Retrieved from https://silo.tips/download/sampo-hietanen-ceo-its-finland.
- Hill, E.J., Ferris, M., Märtinson, V., 2003. Does it matter where you work? A comparison of how three work venues (traditional office, virtual office, and home office) influence aspects of work and personal/family life. J. Vocat. Behav. 63 (2), 220–241. https://doi.org/10.1016/S0001-8791(03)00042-3.
- Hollingsworth, J., Copeland, B., Johnson, J.X., 2019. Are e-scooters polluters? The environmental impacts of shared dockless electric scooters. Environ. Res. Lett. 14 (8), 084031.
- Holmner, Å., Ebi, K.L., Lazuardi, L., Nilsson, M., 2014. Carbon footprint of telemedicine solutions unexplored opportunity for reducing carbon emissions in the health sector. PLoS ONE 9 (9), e105040. https://doi.org/10.1371/journal.pone.0105040.
- Hurlbut, A.R., 2018. Online vs. traditional learning in teacher education: a comparison of student progress. Am. J. Dist. Educ. 32 (4), 248–266. https://doi.org/ 10.1080/08923647.2018.1509265.
- Høyer, K.G., Næss, P., 2001. Conference tourism: a problem for the environment, as well as for research? J. Sustain. Tourism 9 (6), 451–470. https://doi.org/10.1080/ 09669580108667414.
- Jamal, S., Habib, M.A., 2020. Smartphone and daily travel: How the use of smartphone applications affect travel decisions. Sustain. Cities Soc. 53, 101939. https://doi.org/10.1016/j.scs.2019.101939.
- James, O., Swiderski, I.J., Hicks, J., Teoman, D., Buehler, R., 2019. Pedestrians and E-scooters: an initial look at E-scooter parking and perceptions by riders and nonriders. Sustainability 11 (20). https://doi.org/10.3390/su11205591.
- Jia, Y., Fu, H., 2019. Association between innovative dockless bicycle sharing programs and adopting cycling in commuting and non-commuting trips. Transport. Res. Part A: Policy Pract. 121, 12–21. https://doi.org/10.1016/j.tra.2018.12.025.
- Jiao, J., Bai, S., 2020. Understanding the shared e-scooter travels in Austin, TX. ISPRS Int. J. Geo-Inform. 9 (2), 135.
- Jiao, J., Bischak, C., Hyden, S., 2020. The impact of shared mobility on trip generation behavior in the US: findings from the 2017 National Household Travel Survey. Travel Behav. Soc. 19, 1–7. https://doi.org/10.1016/j.tbs.2019.11.001.
- Jin, S.T., Kong, H., Wu, R., Sui, D.Z., 2018. Ridesourcing, the sharing economy, and the future of cities. Cities 76, 96–104. https://doi.org/10.1016/j. cities.2018.01.012.
- Jittrapirom, P., Marchau, V., van der Heijden, R., Meurs, H., 2018. Future implementation of mobility as a service (MaaS): results of an international Delphi study. Travel Behav. Soc. https://doi.org/10.1016/j.tbs.2018.12.004.
- Kamargianni, M., Li, W., Matyas, M., Schäfer, A., 2016. A critical review of new mobility services for urban transport. Transp. Res. Procedia 14, 3294–3303. https:// doi.org/10.1016/j.trpro.2016.05.277.
- Kazekami, S., 2020. Mechanisms to improve labor productivity by performing telework. Telecommun. Policy 44 (2), 101868. https://doi.org/10.1016/j. telpol.2019.101868.
- Kent, J.L., 2014. Carsharing as active transport: what are the potential health benefits? J. Transp. Health 1 (1), 54–62. https://doi.org/10.1016/j.jth.2013.07.003.
- Kent, J.L., Dowling, R., 2013. Puncturing automobility? Carsharing practices. J. Transp. Geogr. 32, 86–92. https://doi.org/10.1016/j.jtrangeo.2013.08.014.
 Kent, J.L., Dowling, R., 2016. "Over 1000 Cars and No Garage": how urban planning supports car(Park) sharing. Urban Policy Res. 34 (3), 256–268. https://doi.org/ 10.1080/08111146.2015.1077806.
- Kenyon, S., Lyons, G., 2007. Introducing multitasking to the study of travel and ICT: examining its extent and assessing its potential importance. Transport. Res. Part A: Policy Pract. 41 (2), 161–175. https://doi.org/10.1016/j.tra.2006.02.004.

- Kim, S.-N., 2017. Is telecommuting sustainable? An alternative approach to estimating the impact of home-based telecommuting on household travel. Int. J. Sustain. Transport. 11 (2), 72–85. https://doi.org/10.1080/15568318.2016.1193779.
- Kim, S.-N., Mokhtarian, P.L., Ahn, K.-H., 2012. The seoul of alonso: new perspectives on telecommuting and residential location from South Korea. Urban Geography 33 (8), 1163–1191. https://doi.org/10.2747/0272-3638.33.8.1163.
- Kim, S.H., Mokhtarian, P.L., Circella, G., 2020a. How, and for whom, will activity patterns be modified by self-driving cars? Expectations from the state of Georgia. Transport. Res. Part F: Traffic Psychol. Behav. 70, 68–80. https://doi.org/10.1016/j.trf.2020.02.012.
- Kim, S.H., Mokhtarian, P.L., Circella, G., 2020b. Will autonomous vehicles change residential location and vehicle ownership? Glimpses from Georgia. Transport. Res. Part D: Transp. Environ. 82, 102291. https://doi.org/10.1016/j.trd.2020.102291.
- Kwan, M.-P., 2007. Mobile communications, social networks, and urban travel: hypertext as a new metaphor for conceptualizing spatial interaction. Profess.
- Geographer 59 (4), 434–446. https://doi.org/10.1111/j.1467-9272.2007.00633.x. Lee, R.J., Sener, I.N., Mokhtarian, P.L., Handy, S.L., 2017. Relationships between the online and in-store shopping frequency of Davis, California residents. Transport. Res. Part A: Policy Pract. 100, 40–52. https://doi.org/10.1016/j.tra.2017.03.001.
- Levinson, D.M., Krizek, K.J., 2017. The End of Traffic and the Future of Access: A Roadmap to the New Transport Landscape. Network Design Lab, Sydney.
- Li, C., Mirosa, M., Bremer, P., 2020. Review of online food delivery platforms and their impacts on sustainability. Sustainability 12 (14). https://doi.org/10.3390/ su12145528.

Li, Y., Voege, T., 2017. Mobility as a service (MaaS): challenges of implementation and policy required. J. Transport. Technol. 7 (02), 95–106.

- Lidynia, C., Philipsen, R., & Ziefle, M., 2017. Droning on About Drones—Acceptance of and Perceived Barriers to Drones in Civil Usage Contexts. Paper presented at the Advances in Human Factors in Robots and Unmanned Systems, Cham.
- Line, T., Jain, J., Lyons, G., 2011. The role of ICTs in everyday mobile lives. J. Transp. Geogr. 19 (6), 1490–1499. https://doi.org/10.1016/j.jtrangeo.2010.07.002. Litman, T., 2000. Evaluating carsharing benefits. Transp. Res. Rec. 1702 (1), 31–35.
- Litman, T., 2020. Autonomous Vehicle Implementation Predictions: Implications for Transport Planning. Victoria Transport Policy Institute, Victoria, Canada Lobato, R., 2019. Netflix Nations: The Geography of Digital Distribution. NYU Press, New York, NY.
- Long, L.N., Hanford, S.D., Janrathitikarn, O., Sinsley, G.L., & Miller, J.A., 2007. A Review of Intelligent Systems Software for Autonomous Vehicles.
- Lyons, G., 2018. Getting smart about urban mobility aligning the paradigms of smart and sustainable. Transport. Res. Part A: Policy Pract. 115, 4–14. https://doi.org/10.1016/j.tra.2016.12.001.
- Lyons, G., Hammond, P., Mackay, K., 2019. The importance of user perspective in the evolution of MaaS. Transport. Res. Part A: Policy Pract. 121, 22–36. https://doi.org/10.1016/j.tra.2018.12.010.
- Lyons, G., Mokhtarian, P., Dijst, M., Böcker, L., 2018. The dynamics of urban metabolism in the face of digitalization and changing lifestyles: understanding and influencing our cities. Resour. Conserv. Recycl. 132, 246–257. https://doi.org/10.1016/j.resconrec.2017.07.032.
- Machado, A.S.C., De Salles Hue, P.N., Berssaneti, T.F., Quintanilha, A.J., 2018. An overview of shared mobility. Sustainability 10 (12). https://doi.org/10.3390/ su10124342.
- Martin, E., Shaheen, S.A., 2011. Greenhouse gas emission impacts of carsharing in North America. IEEE Trans. Intell. Transp. Syst. 12 (4), 1074–1086.
- Martin, E., Shaheen, S.A., Lidicker, J., 2010. Impact of carsharing on household vehicle holdings: results from north american shared-use vehicle survey. Transp. Res. Rec. 2143 (1), 150–158. https://doi.org/10.3141/2143-19.
- McCutcheon, K., Lohan, M., Traynor, M., Martin, D., 2015. A systematic review evaluating the impact of online or blended learning vs. face-to-face learning of clinical skills in undergraduate nurse education. J. Adv. Nurs. 71 (2), 255–270. https://doi.org/10.1111/jan.12509.
- McKenzie, G., 2019. Spatiotemporal comparative analysis of scooter-share and bike-share usage patterns in Washington, D.C. J. Transp. Geogr. 78, 19–28. https://doi.org/10.1016/j.jtrangeo.2019.05.007.
- Meyer, J., Becker, H., Bösch, P.M., Axhausen, K.W., 2017. Autonomous vehicles: the next jump in accessibilities? Res. Transport. Econ. 62, 80–91. https://doi.org/ 10.1016/j.retrec.2017.03.005.
- Milakis, D., Kroesen, M., van Wee, B., 2018. Implications of automated vehicles for accessibility and location choices: evidence from an expert-based experiment. J. Transp. Geogr. 68, 142–148. https://doi.org/10.1016/j.jtrangeo.2018.03.010.
- Milakis, D., van Arem, B., van Wee, B., 2017. Policy and society related implications of automated driving: a review of literature and directions for future research. J. Intell. Transport. Syst. 21 (4), 324–348. https://doi.org/10.1080/15472450.2017.1291351.
- Mohamed, M.J., Rye, T., Fonzone, A., 2019. Operational and policy implications of ridesourcing services: a case of Uber in London, UK. Case Stud. Transp. Policy 7 (4), 823–836. https://doi.org/10.1016/j.cstp.2019.07.013.
- Mokhtarian, P.L., 2004. A conceptual analysis of the transportation impacts of B2C e-commerce. Transportation 31 (3), 257–284. https://doi.org/10.1023/B: PORT.0000025428.64128.d3.
- Mokhtarian, P.L., Chen, C., 2004. TTB or not TTB, that is the question: a review and analysis of the empirical literature on travel time (and money) budgets. Transport. Res. Part A: Policy Pract. 38 (9), 643–675. https://doi.org/10.1016/j.tra.2003.12.004.
- Mokhtarian, P.L., Salomon, I., Handy, S.L., 2006. The impacts of ICT on leisure activities and travel: a conceptual exploration. Transportation 33 (3), 263–289. https://doi.org/10.1007/s11116-005-2305-6.
- Moriset, B., 2003. The new economy in the city: emergence and location factors of internet-based companies in the metropolitan area of Lyon, France. Urban Stud. 40 (11), 2165–2186. https://doi.org/10.1080/0042098032000123231.
- Moscholidou, I., Pangbourne, K., 2019. A preliminary assessment of regulatory efforts to steer smart mobility in London and Seattle. Transp. Policy. https://doi.org/ 10.1016/i.tranpol.2019.10.015.
- Mouratidis, K., Cobeña Serrano, V., 2021. Autonomous buses: Intentions to use, passenger experiences, and suggestions for improvement. Transport. Res. Part F: Traff. Psychol. Behav. 76, 321–335. https://doi.org/10.1016/j.trf.2020.12.007.
- Mouratidis, K., Ettema, D., Næss, P., 2019. Urban form, travel behavior, and travel satisfaction. Transport. Res. Part A: Policy Pract. 129, 306–320. https://doi.org/ 10.1016/j.tra.2019.09.002.
- Murray, C.C., Chu, A.G., 2015. The flying sidekick traveling salesman problem: optimization of drone-assisted parcel delivery. Transport. Res. Part C: Emerg. Technol. 54, 86–109. https://doi.org/10.1016/j.trc.2015.03.005.
- Narayanan, S., Chaniotakis, E., Antoniou, C., 2020. Shared autonomous vehicle services: a comprehensive review. Transport. Res. Part C: Emerg. Technol. 111, 255–293. https://doi.org/10.1016/j.trc.2019.12.008.
- NASA, 2018. Urban Air Mobility (UAM) Market Study. Retrieved from Washington, D.C. https://www.nasa.gov/sites/default/files/atoms/files/uam-market-studyexecutive-summary-v2.pdf.
- Nazari, F., Noruzoliaee, M., Mohammadian, A., 2018. Shared versus private mobility: modeling public interest in autonomous vehicles accounting for latent attitudes. Transport. Res. Part C: Emerg. Technol. 97, 456–477. https://doi.org/10.1016/j.trc.2018.11.005.
- Nenseth, V., Ciccone, A., & Kristensen, N.B., 2019. Societal consequences of automated vehicles: Norwegian scenarios (8248022374). Retrieved from Oslo: https://www.toi.no/getfile.php?mmfileid=50576.
- Newton, P.W., 2012. Liveable and sustainable? Socio-technical challenges for twenty-first-century cities. J. Urban Technol. 19 (1), 81–102. https://doi.org/10.1080/10630732.2012.626703.
- Nordhoff, S., de Winter, J., Madigan, R., Merat, N., van Arem, B., Happee, R., 2018. User acceptance of automated shuttles in Berlin-Schöneberg: a questionnaire study. Transport. Res. Part F: Traffic Psychol. Behav. 58, 843–854. https://doi.org/10.1016/j.trf.2018.06.024.
- Ory, D.T., Mokhtarian, P.L., 2006. Which came first, the telecommuting or the residential relocation? An empirical analysis of causality. Urban Geography 27 (7), 590–609. https://doi.org/10.2747/0272-3638.27.7.590.
- Pangbourne, K., Mladenović, M.N., Stead, D., Milakis, D., 2019. Questioning mobility as a service: unanticipated implications for society and governance. Transport. Res. Part A: Policy Pract. https://doi.org/10.1016/j.tra.2019.09.033.

- Pawlak, J., 2020. Travel-based multitasking: review of the role of digital activities and connectivity. Transp. Rev. 40 (4), 429–456. https://doi.org/10.1080/01441647.2020.1728418.
- Pawlak, J., Circella, G., Mahmassani, H., & Mokhtarian, P. (2019). Information and Communication Technologies (ICT), Activity Decisions, and Travel Choices: 20 years into the Second Millennium and where do we go next? Retrieved from Washington, DC: http://onlinepubs.trb.org/onlinepubs/centennial/papers/ADB20-Final.pdf.
- Pérez Pérez, M., 2004. The environmental impacts of teleworking: a model of urban analysis and a case study. Manage. Environ. Qual.: Int. J. 15 (6), 656–671. https://doi.org/10.1108/14777830410560728.
- Pinto, R., Zambetti, M., Lagorio, A., Pirola, F., 2019. A network design model for a meal delivery service using drones. Int. J. Logist. Res. Appl. 1–21 https://doi.org/ 10.1080/13675567.2019.1696290.
- Pratt, J.H., 2000. Asking the right questions about telecommuting: avoiding pitfalls in surveying homebased work. Transportation 27 (1), 99–116. https://doi.org/ 10.1023/A:1005288112292.
- Rahim Taleqani, A., Hough, J., Nygard, K.E., 2019. Public opinion on dockless bike sharing: a machine learning approach. Transp. Res. Rec. 2673 (4), 195–204. https://doi.org/10.1177/0361198119838982.
- Rayle, L., Dai, D., Chan, N., Cervero, R., Shaheen, S., 2016. Just a better taxi? A survey-based comparison of taxis, transit, and ridesourcing services in San Francisco. Transp. Policy 45, 168–178. https://doi.org/10.1016/j.tranpol.2015.10.004.
- Raza, A., Ali, S., Akram, M., 2019. Immunity-based dynamic reconfiguration of mobile robots in unstructured environments. J. Intell. Rob. Syst. 96 (3), 501–515. https://doi.org/10.1007/s10846-019-01000-6.
- Rosenfeld, M.J., Thomas, R.J., Hausen, S., 2019. Disintermediating your friends: How online dating in the United States displaces other ways of meeting. Proc. Natl. Acad. Sci. 116 (36), 17753. https://doi.org/10.1073/pnas.1908630116.
- Rotaris, L., Danielis, R., 2018. The role for carsharing in medium to small-sized towns and in less-densely populated rural areas. Transport. Res. Part A: Policy Pract. 115, 49–62. https://doi.org/10.1016/j.tra.2017.07.006.
- Rotem-Mindali, O., Salomon, I., 2007. The impacts of E-retail on the choice of shopping trips and delivery: some preliminary findings. Transport. Res. Part A: Policy Pract. 41 (2), 176–189. https://doi.org/10.1016/j.tra.2006.02.007.
- Roy, R., Potter, S., Yarrow, K., 2008. Designing low carbon higher education systems: environmental impacts of campus and distance learning systems. Int. J. Sustain. High. Educ. 9 (2), 116–130. https://doi.org/10.1108/14676370810856279.
- Salomon, I., 1986. Telecommunications and travel relationships: a review. Transport. Res. Part A: Gen. 20 (3), 223-238. https://doi.org/10.1016/0191-2607(86) 90096-8.
- Salonen, A.O., Haavisto, N., 2019. Towards autonomous transportation. Passengers' experiences, perceptions and feelings in a driverless shuttle bus in Finland. Sustainability 11 (3). https://doi.org/10.3390/su11030588.
- Serrao, J., Nilsson, S., & Kimmel, S., 2018. A Legal and Regulatory Assessment for the Potential of Urban Air Mobility (UAM). Retrieved from UC Berkeley: https://escholarship.org/uc/item/49b8b9w0.
- Shaheen, S., Chan, N., Gaynor, T., 2016a. Casual carpooling in the San Francisco Bay Area: understanding user characteristics, behaviors, and motivations. Transp. Policy 51, 165–173. https://doi.org/10.1016/j.tranpol.2016.01.003.
- Shaheen, S., Cohen, A., 2013. Carsharing and personal vehicle services: worldwide market developments and emerging trends. Int. J. Sustain. Transport. 7 (1), 5–34. https://doi.org/10.1080/15568318.2012.660103.
- Shaheen, S., Cohen, A., 2019. Shared ride services in North America: definitions, impacts, and the future of pooling. Transp. Rev. 39 (4), 427–442. https://doi.org/ 10.1080/01441647.2018.1497728.
- Shaheen, S., Cohen, A., & Farrar, E., 2018. The potential societal barriers of urban air mobility (UAM). National Aeronautics and Space Administration (NASA). Retrieved from UC Berkeley: https://escholarship.org/uc/item/7p69d2bg.
- Shaheen, S., Cohen, A., Farrar, E., 2019. Chapter five Carsharing's impact and future. In: Fishman, E. (Ed.), Advances in Transport Policy and Planning, Vol. 4. Academic Press, Cambridge, MA, pp. 87–120.
- Shaheen, S., Cohen, A., Yelchuru, B., Sarkhili, S., & Hamilton, B.A., 2017. Mobility on demand operational concept report. Retrieved from https://rosap.ntl.bts.gov/ view/dot/34258.
- Shaheen, S., Cohen, A., & Zohdy, I., 2016. Shared mobility: current practices and guiding principles. Retrieved from https://ops.fhwa.dot.gov/publications/ fhwahop16022/fhwahop16022.pdf.
- Shaheen, S., Guzman, S., Zhang, H., 2010. Bikesharing in Europe, the Americas, and Asia: past, present, and future. Transp. Res. Rec. 2143 (1), 159–167. https://doi.org/10.3141/2143-20.
- Sioui, L., Morency, C., Trépanier, M., 2013. How carsharing affects the travel behavior of households: a case study of Montréal, Canada. Int. J. Sustain. Transport. 7 (1), 52–69. https://doi.org/10.1080/15568318.2012.660109.
- Smith, A.C., Thomas, E., Snoswell, C.L., Haydon, H., Mehrotra, A., Clemensen, J., Caffery, L.J., 2020. Telehealth for global emergencies: implications for coronavirus disease 2019 (COVID-19). J. Telemed. Telecare 26 (5), 309–313. https://doi.org/10.1177/1357633X20916567.
- Sochor, J., Arby, H., Karlsson, I.C.M., Sarasini, S., 2018. A topological approach to Mobility as a Service: a proposed tool for understanding requirements and effects, and for aiding the integration of societal goals. Res. Transport. Bus. Manage. 27, 3–14. https://doi.org/10.1016/j.rtbm.2018.12.003.
- Soteropoulos, A., Berger, M., Ciari, F., 2019. Impacts of automated vehicles on travel behaviour and land use: an international review of modelling studies. Transp. Rev. 39 (1), 29–49. https://doi.org/10.1080/01441647.2018.1523253.
- Sperling, D., 2018. Three Revolutions: Steering Automated, Shared, and Electric Vehicles to a Better Future. Island Press, Washington, DC.
- Standing, C., Standing, S., Biermann, S., 2019. The implications of the sharing economy for transport. Transp. Rev. 39 (2), 226–242. https://doi.org/10.1080/ 01441647.2018.1450307.
- Stead, D., Vaddadi, B., 2019. Automated vehicles and how they may affect urban form: a review of recent scenario studies. Cities 92, 125–133. https://doi.org/ 10.1016/j.cities.2019.03.020.
- Stillwater, T., Mokhtarian, P.L., Shaheen, S.A., 2009. Carsharing and the built environment: geographic information system-based study of one U.S. Operator. Transport. Res. Rec. 2110 (1), 27–34. https://doi.org/10.3141/2110-04.
- Summers, J.J., Waigandt, A., Whittaker, T.A., 2005. A comparison of student achievement and satisfaction in an online versus a traditional face-to-face statistics class. Innovat. Higher Educ. 29 (3), 233–250.
- Taeihagh, A., Lim, H.S.M., 2019. Governing autonomous vehicles: emerging responses for safety, liability, privacy, cybersecurity, and industry risks. Transp. Rev. 39 (1), 103–128. https://doi.org/10.1080/01441647.2018.1494640.
- Tarafdar, M., Maier, C., Laumer, S., Weitzel, T., 2020. Explaining the link between technostress and technology addiction for social networking sites: a study of distraction as a coping behavior. Inform. Syst. J. 30 (1), 96–124. https://doi.org/10.1111/isj.12253.
- Tayyaran, M.R., Khan, A.M., 2003. The effects of telecommuting and intelligent transportation systems on urban development. J. Urban Technol. 10 (2), 87–100. https://doi.org/10.1080/1063073032000139714.
- Tayyaran, M.R., Khan, A.M., Anderson, D.A., 2003. Impact of telecommuting and intelligent transportation systems on residential location choice. Transport. Plan. Technol. 26 (2), 171–193. https://doi.org/10.1080/715020598.
- Tirachini, A., 2020. Ride-hailing, travel behaviour and sustainable mobility: an international review. Transportation 47 (4), 2011–2047. https://doi.org/10.1007/s11116-019-10070-2.
- Tirachini, A., Antoniou, C., 2020. The economics of automated public transport: effects on operator cost, travel time, fare and subsidy. Econ. Transport. 21, 100151. https://doi.org/10.1016/j.ecotra.2019.100151.
- Tirachini, A., del Río, M., 2019. Ride-hailing in Santiago de Chile: users' characterisation and effects on travel behaviour. Transp. Policy 82, 46–57. https://doi.org/ 10.1016/j.tranpol.2019.07.008.

Tussyadiah, I.P., Pesonen, J., 2015. Impacts of peer-to-peer accommodation use on travel patterns. J. Travel Res. 55 (8), 1022–1040. https://doi.org/10.1177/0047287515608505.

Ulmer, M.W., Thomas, B.W., 2018. Same-day delivery with heterogeneous fleets of drones and vehicles. Networks 72 (4), 475–505. https://doi.org/10.1002/net.21855.

Uteng, T.P., Christensen, H.R., Levin, L., 2020. Gendering Smart Mobilities. Routledge, London.

Utriainen, R., Pöllänen, M., 2018. Review on mobility as a service in scientific publications. Res. Transport. Bus. Manage. 27, 15–23. https://doi.org/10.1016/j. rtbm.2018.10.005.

van Wee, B., Banister, D., 2016. How to write a literature review paper? Transp. Rev. 36 (2), 278-288. https://doi.org/10.1080/01441647.2015.1065456.

van Wee, B., Geurs, K., Chorus, C., 2013. Information, communication, travel behavior and accessibility. J. Transp. Land Use 6 (3), 1–16. https://doi.org/10.5198/ jtlu.v6i3.282.

Varghese, V., Jana, A., 2019. Interrelationships between ICT, social disadvantage, and activity participation behaviour: a case of Mumbai, India. Transport. Res. Part A: Policy Pract. 125, 248–267. https://doi.org/10.1016/j.tra.2018.06.009.

Vilhelmson, B., Thulin, E., 2016. Who and where are the flexible workers? Exploring the current diffusion of telework in Sweden. New Technol. Work Empl. 31 (1), 77–96. https://doi.org/10.1111/ntwe.12060.

Voytenko Palgan, Y., Zvoľska, L., Mont, O., 2017. Sustainability framings of accommodation sharing. Environ. Innovat. Trans. 23, 70–83. https://doi.org/10.1016/j.eist.2016.12.002.

Wang, D., Law, F.Y.T., 2007. Impacts of Information and Communication Technologies (ICT) on time use and travel behavior: a structural equations analysis. Transportation 34 (4), 513–527. https://doi.org/10.1007/s11116-007-9113-0.

Wang, J., Lindsey, G., 2019. Do new bike share stations increase member use: A quasi-experimental study. Transport. Res. Part A: Policy Pract. 121, 1–11. https://doi.org/10.1016/j.tra.2019.01.004.

Wenninger, H., Krasnova, H., Buxmann, P., 2019. Understanding the role of social networking sites in the subjective well-being of users: a diary study. Eur. J. Inform. Syst. 28 (2), 126–148. https://doi.org/10.1080/0960085X.2018.1496883.

Wijesooriya, N.R., Mishra, V., Brand, P.L.P., Rubin, B.K., 2020. COVID-19 and telehealth, education, and research adaptations. Paediatr. Respir. Rev. 35, 38-42. https://doi.org/10.1016/j.prrv.2020.06.009.

Wong, Y.Z., Hensher, D.A., Mulley, C., 2019. Mobility as a service (MaaS): charting a future context. Transport. Res. Part A: Policy Pract. https://doi.org/10.1016/j. tra.2019.09.030.

World Economic Forum, 2020. The Future of the Last-Mile Ecosystem. Retrieved from Geneva: http://www3.weforum.org/docs/WEF_Future_of_the_last_mile_ecosystem.pdf.

Xanthidis, D., Alali, A.S., Koutzampasopoulou, O., 2016. Online socializing: how does it affect the information seeking behavior and the educational preferences in Saudi Arabia? Comput. Hum. Behav. 60, 425–434. https://doi.org/10.1016/j.chb.2016.02.062.

Xi, G., Cao, X., Zhen, F., 2020. The impacts of same day delivery online shopping on local store shopping in Nanjing, China. Transport. Res. Part A: Policy Pract. 136, 35–47. https://doi.org/10.1016/j.tra.2020.03.030.

Xi, G., Zhen, F., Cao, X., Xu, F., 2018. The interaction between e-shopping and store shopping: empirical evidence from Nanjing, China. Transport. Lett. 1–9 https:// doi.org/10.1080/19427867.2018.1546797.

Xu, D., Jaggars, S.S., 2013. The impact of online learning on students' course outcomes: evidence from a large community and technical college system. Econ. Educ. Rev. 37, 46–57. https://doi.org/10.1016/j.econedurev.2013.08.001.

Young, M., Farber, S., 2019. The who, why, and when of Uber and other ride-hailing trips: an examination of a large sample household travel survey. Transport. Res. Part A: Policy Pract. 119, 383-392. https://doi.org/10.1016/j.tra.2018.11.018.

Yousefi, Z., Dadashpoor, H., 2019. How do ICTs affect urban spatial structure? A systematic literature review. J. Urban Technol. 1–19 https://doi.org/10.1080/ 10630732.2019.1689593.

Zakharenko, R., 2016. Self-driving cars will change cities. Reg. Sci. Urban Econ. 26-37. https://doi.org/10.1016/j.regsciurbeco.2016.09.003.

Zhang, W., Guhathakurta, S., 2018. Residential location choice in the era of shared autonomous vehicles. J. Plan. Educ. Res. 0739456X18776062 https://doi.org/ 10.1177/0739456X18776062.

Zhang, W., Wang, K., 2020. Parking futures: Shared automated vehicles and parking demand reduction trajectories in Atlanta. Land Use Policy 91, 103963. https://doi.org/10.1016/j.landusepol.2019.04.024.

Zhen, F., Cao, X., Mokhtarian, P.L., Xi, G., 2016. Associations between online purchasing and store purchasing for four types of products in Nanjing, China. Transport. Res. Rec. 2566 (1), 93–101. https://doi.org/10.3141/2566-10.

Zhen, F., Du, X., Cao, J., Mokhtarian, P.L., 2018. The association between spatial attributes and e-shopping in the shopping process for search goods and experience goods: evidence from Nanjing. J. Transp. Geogr. 66, 291–299. https://doi.org/10.1016/j.jtrangeo.2017.11.007.

Zhou, Y., Wang, X., 2014. Explore the relationship between online shopping and shopping trips: an analysis with the 2009 NHTS data. Transport. Res. Part A: Policy Pract. 70, 1–9. https://doi.org/10.1016/j.tra.2014.09.014.

Zhu, P., 2012. Are telecommuting and personal travel complements or substitutes? Ann. Reg. Sci. 48 (2), 619–639. https://doi.org/10.1007/s00168-011-0460-6. Zhuang, D., Jin, J.G., Shen, Y., Jiang, W., 2019. Understanding the bike sharing travel demand and cycle lane network: the case of Shanghai. Int. J. Sustain. Transport.

1-13 https://doi.org/10.1080/15568318.2019.1699209.